

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Lachowska, Marta; Mas, Alexandre; Woodbury, Stephen A.

## Working Paper Sources of displaced workers' long-term earnings losses

Upjohn Institute Working Paper, No. 18-282

**Provided in Cooperation with:** W. E. Upjohn Institute for Employment Research, Kalamazoo, Mich.

*Suggested Citation:* Lachowska, Marta; Mas, Alexandre; Woodbury, Stephen A. (2017) : Sources of displaced workers' long-term earnings losses, Upjohn Institute Working Paper, No. 18-282, W.E. Upjohn Institute for Employment Research, Kalamazoo, MI, https://doi.org/10.17848/wp18-282

This Version is available at: https://hdl.handle.net/10419/202880

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU

# UPJOHN INSTITUTE

Upjohn Institute Working Papers

Upjohn Research home page

2018

# Sources of Displaced Workers' Long-Term Earnings Losses

Marta Lachowska W.E. Upjohn Institute, marta@upjohn.org

Alexandre Mas Princeton University

Stephen A. Woodbury Michigan State University and W.E. Upjohn Institute, woodbury@upjohn.org

Upjohn Institute working paper; 18-282

Citation

Lachowska, Marta, Alexandre Mas, Stephen A. Woodbury. 2018. "Sources of Displaced Workers' Long-Term Earnings Losses." Upjohn Institute Working Paper 18-282. Kalamazoo, MI: Upjohn Institute for Employment Research. https://doi.org/10.17848/ wp18-282

This title is brought to you by the Upjohn Institute. For more information, please contact ir@upjohn.org.

W.E. UPJOHN INSTITUTE FOR EMPLOYMENT RESEARCH

# SOURCES OF DISPLACED WORKERS'

Researching the causes and consequences of unemployment

## LONG-TERM EARNINGS LOSSES

#### **Upjohn Institute Working Paper 18-282**

Marta Lachowska W.E. Upjohn Institute for Employment Research email: marta@upjohn.org

> Alexandre Mas Princeton University and NBER

Stephen A. Woodbury Michigan State University and W.E. Upjohn Institute

December 31, 2017

#### ABSTRACT

We estimate the earnings losses of a cohort of workers displaced during the Great Recession and decompose those long-term losses into components attributable to fewer work hours and to reduced hourly wage rates. We also examine the extent to which the reduced earnings, work hours, and wages of these displaced workers can be attributed to factors specific to pre- and post-displacement employers; that is, to employer-specific fixed effects. The analysis is based on employer-employee linked panel data from Washington State assembled from 2002–2014 administrative wage and unemployment insurance (UI) records.

Three main findings emerge from the empirical work. First, five years after job loss, the earnings of these displaced workers were 16 percent less than those of comparison groups of non-displaced workers. Second, earnings losses within a year of displacement can be explained almost entirely by lost work hours; however, five years after displacement, the relative earnings deficit of displaced workers can be attributed roughly 40 percent to reduced hourly wages and 60 percent to reduce work hours. Third for the average displaced worker, lost employer-specific premiums account for about 11 percent of long-term earnings losses and nearly 25 percent of lower long-term hourly wages. For workers displaced from employers paying top-quintile earnings premiums (about 60 percent of the displaced workers in the sample), lost employer-specific premiums account for more than half of long-term earnings losses and 83 percent of lower long-term hourly wages.

#### JEL Classification Codes: C21, J22, J23, J38, J65

**Key Words**: Earnings, hours, wages, displaced workers, unemployment, reemployment policy, administrative data, difference-in-differences, employer- vs. individual-specific earnings and wage differentials

**Acknowledgments**: For comments on an earlier draft, we are grateful to Katharine Abraham, David Card, Henry Farber, Patrick Kline, Pawel Krolikowski, Peter Kuhn, Matthew Notowidigdo, Jesse Rothstein, Isaac Sorkin, Desmond Toohey, and participants in seminars at Delaware, IZA, SOLE, and the NBER Summer Institute. We gratefully acknowledge the financial support of the Washington Center for Equitable Growth and the Russell Sage Foundation.

Upjohn Institute working papers are meant to stimulate discussion and criticism among the policy research community. Content and opinions are the sole responsibility of the author.

### Sources of Displaced Workers' Long-Term Earnings Losses<sup>1</sup>

Marta Lachowska<sup>2</sup> Alexandre Mas<sup>3</sup> Stephen A. Woodbury<sup>4</sup> December 31, 2017

#### Abstract

We estimate the earnings losses of a cohort of workers displaced during the Great Recession and decompose those long-term losses into components attributable to fewer work hours and to reduced hourly wage rates. We also examine the extent to which the reduced earnings, work hours, and wages of these displaced workers can be attributed to factors specific to pre- and post-displacement employers; that is, to employer-specific fixed effects. The analysis is based on employer-employee linked panel data from Washington State assembled from 2002–2014 administrative wage and unemployment insurance (UI) records.

Three main findings emerge from the empirical work. First, five years after job loss, the earnings of these displaced workers were 16 percent less than those of comparison groups of nondisplaced workers. Second, earnings losses within a year of displacement can be explained almost entirely by lost work hours; however, five years after displacement, the relative earnings deficit of displaced workers can be attributed roughly 40 percent to reduced hourly wages and 60 percent to reduced work hours. Third, for the average displaced worker, lost employer-specific premiums account for about 11 percent of long-term earnings losses and nearly 25 percent of lower long-term hourly wages. For workers displaced from employers paying top-quintile earnings premiums (about 60 percent of the displaced workers in the sample), lost employer-specific premiums account for more than half of long-term earnings losses and 83 percent of lower long-term hourly wages.

#### JEL classifications: C21, J22, J23, J38, J65

*Keywords*: Earnings, Hours, Wages, Displaced workers, Unemployment, Reemployment policy, Administrative data, Difference-in-differences, Employer- vs. individual-specific earnings and wage differentials

<sup>&</sup>lt;sup>1</sup> For comments on an earlier draft, we are grateful to Katharine Abraham, David Card, Henry Farber, Patrick Kline, Pawel Krolikowski, Peter Kuhn, Matthew Notowidigdo, Jesse Rothstein, Isaac Sorkin, Desmond Toohey, and participants in seminars at Delaware, IZA, SOLE, and the NBER Summer Institute. We gratefully acknowledge the financial support of the Washington Center for Equitable Growth and the Russell Sage Foundation.

<sup>&</sup>lt;sup>2</sup> W.E. Upjohn Institute for Employment Research, 300 South Westnedge Avenue, Kalamazoo, MI 49007, United States (marta@upjohn.org).

<sup>&</sup>lt;sup>3</sup> Industrial Relations Section and Department of Economics, Princeton University (mas@princeton.edu), and NBER.

<sup>&</sup>lt;sup>4</sup> Department of Economics, Michigan State University (woodbur2@msu.edu), and W.E. Upjohn Institute.

#### **1** Introduction

It is well-known that permanent loss of a long-term job—worker displacement—leads to earnings losses that are enduring, even permanent.<sup>5</sup> But long-standing unanswered questions surround the reasons for these losses. Are they the result of lost specific human capital, lost firmspecific rents, unobserved characteristics of displaced workers, or something else? Longitudinal data on workers' earnings have established the magnitude of displaced workers' earnings losses, but they have not been able to answer questions about the reasons for these losses, which are important both theoretically and from the standpoint of mitigating the losses.

In this paper, we use linked employer-employee panel data based on administrative records from the unemployment insurance (UI) system of Washington State (both wage records and UI claims records from 2002–2014) to obtain evidence on the sources of dislocated workers' earnings losses. The Washington administrative records are unusual because, in addition to reporting employer-specific quarterly earnings of all UI-covered workers in the state, they also report quarterly work hours. The availability of both earnings and work hours allows calculation of hourly wage rates on a quarterly basis; this in turn allows us to decompose displaced workers' long-term earnings losses into components due to reduced hours and lower hourly wages.

The Washington data also give us the opportunity to make headway in understanding the role of employers in generating earnings losses by allowing estimation of worker- and employer-specific fixed effects—the Abowd, Kramarz, and Margolis (1999) (AKM) approach. The

<sup>&</sup>lt;sup>5</sup> See, for example, Topel (1990), Jacobson, LaLonde, and Sullivan (1993a, 1993b), Farber (1993, 1997, 2015, 2017), Stevens (1997), von Wachter, Song, and Manchester (2009), Couch and Placzek (2010), Davis and von Wachter (2012), and the reviews by Hamermesh (1996), Fallick (1996), Kletzer (1998), von Wachter (2010), and Carrington and Fallick (2017). Worker displacement has also been shown to reduce household expenditure (Stephens 2001), lead to poorer health (Schaller and Stevens 2015), reduce happiness (Kalil and DeLeire 2013), increase mortality (Sullivan and von Wachter 2009), and harm children affected by parental job loss (Oreopoulos, Page, and Stevens 2008; Stevens and Schaller 2011).

estimated employer fixed effects allow us to observe whether displaced workers are separating from employers who pay premium earnings and wages, and finding reemployment with employers who do not. Widely accepted interpretations of the employer fixed effects allow us to infer the extent to which earnings losses stem from lost firm-specific rents or, alternatively, lost firm-specific human capital.

We find that workers displaced in Washington during the Great Recession suffered earnings losses similar to those in Pennsylvania during the 1980s (Jacobson, LaLonde, and Sullivan (1993a, 1993b) (hereafter, JLS), in Connecticut during 2000–2001 (Couch and Placzek 2010) (hereafter, CP), and in the U.S. (nationally) over the 1980–2005 period (Davis and von Wachter 2012): five years after job loss, displaced workers' earnings were 16 percent less than those of a stably employed comparison group.<sup>6</sup> The decomposition of these losses into hours and wage rates reveals that virtually all earnings losses in the year following displacement resulted from lost work hours. But the picture changes substantially over time: five years after displacement, about 45 percent of lost earnings were due to reduced work hours, and about 55 percent were due to lower hourly wage rates.<sup>7</sup> Moreover, the pattern of displaced workers' wage rate losses differs markedly from that of earnings losses: whereas earnings follow a well-known pattern of "dip, drop, and partial recovery," wage rates drop suddenly at the time of displacement and recover hardly at all.

Consistent with the recent literature on firm-specific earnings differentials (e.g., Card, Heining, and Kline 2013; Card, Cardoso, and Kline 2016; Sorkin 2016, Barth et al. 2016; Goldschmidt and Schmieder 2017; Fackler et al. 2017), the estimates suggest that employer fixed

<sup>&</sup>lt;sup>6</sup> JLS and CP also use administrative records, so we will draw frequent comparisons between our estimates of displaced workers' earnings losses and theirs.

<sup>&</sup>lt;sup>7</sup> This differs from Stevens's (1997) findings using the Panel Study of Income Dynamics, which suggest that reduced work hours play a relatively minor role in explaining the long-term earnings losses of displaced workers.

effects play a significant role in explaining displaced workers' earnings losses. Specifically, in the long run, employer fixed effects account for about 11 percent of the *average* displaced worker's lost earnings and about 25 percent of their reduced wage rates. For workers displaced from employers paying top-quintile earnings premiums (nearly 60 percent of the displaced workers in the sample), the contribution of lost employer-specific premiums is large and striking: for these workers, more than half of earnings losses and 83 percent of lower long-term hourly wages can be attributed to lost employer-specific premiums. That is, a majority of displaced workers in the sample separated from a high-premium employer, and more than half of the losses of these workers can be attributed the foregone employer-specific fixed effects.

Overall, the findings suggest that, for the average displaced worker, lost firm-specific human capital (or possibly lost match-specific fixed effects) is more important than lost firmspecific rents in explaining lost earnings. In contrast, for workers displaced from employers paying top-quintile earnings premiums, foregone firm-specific rents are central to lost earnings.

The paper is organized as follows. Section 2 describes the data, and section 3 describes the empirical strategy. This starts with JLS's seminal approach, which we extend to a decomposition of earnings losses into components attributable to lost work hours and reduced wage rates. We then combine the JLS approach with the AKM model to examine the importance of employer fixed effects in explaining the losses of displaced workers. Section 4 presents the main results on earnings losses and their decomposition into effects due to lost work hours and lower hourly wage rates. Section 5 examines the importance of employer-specific fixed effects in explaining the losses of displaced of employer-specific fixed effects in explaining the losses of employer-specific fixed effects in explaining the losses of displaced workers. The final section reviews the estimates and discusses their implications.

#### 2 Data

The data we use come from the records maintained by the Employment Security Department of Washington State to administer the state's UI system: quarterly earnings records from all UI-covered employers in Washington from 2002:I through 2014:IV; and the UI claims records of all individuals who claimed UI in Washington at any time during the same period.

The administrative records of most states include only a worker's quarterly earnings by employer; however, UI-covered employers in Washington are required to report each worker's quarterly work hours in addition to quarterly earnings.<sup>8</sup> Hence, a record appears for each quarter-worker-employer combination that includes information on the worker's earnings and work hours during the quarter with the specified employer. This is a distinctive advantage of the Washington administrative data, making it possible to construct an hourly wage rate in quarter *t* for most workers in Washington's formal labor market. Specifically, we rely on the wage rate with the primary employer in each quarter (that is, the employer from whom the worker had the largest share of earnings in a given quarter), dividing earnings from that employer by hours worked with that employer.<sup>9</sup>

The wage records include an employer identifier and the four-digit North American

<sup>&</sup>lt;sup>8</sup> Washington is the only state that uses hours worked in the year before claiming UI to determine UI eligibility, so employers are required to report hours, including overtime hours and hours of paid leave. Actual hours of salaried, commissioned, and piecework employees are reported unless those hours are not tracked, in which case employers are instructed to report 40 hours per week—see *Unemployment Insurance Tax Information*, Employment Security Department, Washington State, October 2014 (Revised). Our examination of the hours data starting in 2001 suggests they are reliable and of high quality—see Lachowska, Mas, and Woodbury (2017). For further discussion of Washington's UI system, see Lachowska, Meral, and Woodbury (2016).

<sup>&</sup>lt;sup>9</sup> All earnings are converted to constant 2010 dollars using the Consumer Price Index for All Urban Consumers (CPI-U). We handle outliers by winsorizing positive earnings at the 99th percentile (about \$69,000 per quarter). Work hours coded "9999" are set to missing. Most observations with positive earnings and zero hours in a quarter are accurate: the Washington Employment Security Department instructs employers to report back pay, bonuses, commissions, cafeteria and 401k plan payments, royalties and residuals, severance and separation pay, settlements, sick leave, and tips and gratuities as quarter *t* earnings if they were paid in quarter *t*, even if the worker no longer worked for the employer in quarter *t* (see https://esd.wa.gov/employer-taxes/zero-hour-reports). Finally, we winsorize positive hours at the 99th percentile (783 hours per quarter) and winsorize positive wage rates at the 99 percentile (about \$139 per hour).

Industry Classification System (NAICS) code of each employer. These make it possible to construct employment at the employer level by summing over the records associated with a given employer in each quarter.

At the worker level, the employer identifiers and NAICS codes, along with the panel nature of the administrative records, allow us to observe worker transitions between employers and industries. The panel nature of the wage records also allows us to observe each worker's tenure with a given employer.

UI administrative records rarely include demographic characteristics. Rather, states typically record worker characteristics only when a worker claims UI benefits and registers for employment services through the public labor exchange. For that reason, we know worker characteristics—age, gender, race, education—only for the subset of workers who claimed UI and registered with the public labor exchange at some point during 2002–2014. For gender and race, we assign an indicator with a constant value throughout the 13-year period. We assign the age of a worker in each quarter based on the worker's age in the quarter he or she was observed. For education, we assign a constant level if we observe the worker only once; however, if we observe the worker more than once (that is, if he or she claimed UI more than once), we assign the first observed value of education for all quarters until the quarter in which we observe a change.

The advantages of the Washington administrative data are substantial—they provide information on the earnings and work hours of virtually all workers employed in Washington over a period of 13 years, as well as information on all UI-covered employers in the state.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup> Exemptions from coverage are limited to the self-employed, including outside sales workers paid solely by commission and independent contractors meeting exemption tests specified in Washington's UI law (Revised Code of Washington, Title 50). Nonprofit religious organizations are also exempt.

Nevertheless, three shortcomings common to UI administrative records are also clear: demographic information is available for only a subsample, the data are restricted to a single state, and information on workers who leave Washington (or leave formal employment for selfemployment or the informal sector) are not available.<sup>11</sup>

We use the Washington administrative records just described for two distinct analyses: an AKM analysis, which estimates employer fixed effects for earnings, work hours, and wage rates using data on all UI-covered workers and employers in Washington; and an analysis of displaced workers' earnings losses, part of which makes use of the AKM analysis. In the rest of this section, we focus on describing the sample used in the displaced worker analysis. The dataset used in the AKM analysis is far larger and is described in Appendix B, along with the AKM analysis itself.

#### 2.1 Construction of the displaced worker analysis sample

We follow JLS and CP in focusing on long-tenure workers, defined as workers with at least six years of job tenure (24 consecutive quarters of positive earnings) with the same primary employer during 2002–2007.<sup>12</sup> Specifically, all workers in the analysis sample (both displaced and non-displaced) have positive earnings or hours with the same primary employer in all quarters from 2002:I–2007:IV.

We define a long-tenure worker as displaced if, at any time during 2008–2010, that worker separated from her primary employer within four quarters of a quarter in which that employer experienced a mass layoff (as defined next). We focus on separations during 2008–

<sup>&</sup>lt;sup>11</sup> As discussed below, this last issue can be mitigated by restricting the analysis sample of the displaced workers to have at least one quarter of positive earnings or work hours in each year starting in 2008. We also check the sensitivity of the estimates to selective attrition (see section 4.5).

<sup>&</sup>lt;sup>12</sup> In section 4.4, we develop estimates for displaced workers with shorter previous job tenures.

2010 because, although the Great Recession contraction officially lasted from December 2007 until June 2009, the recovery of the labor market lagged substantially: the Washington unemployment rate did not fall below 10 percent until June 2010, and had fallen only to 9.6 percent by December 2010 (https://www.bls.gov/lau/). A worker's displacement is dated to the quarter of his or her separation (not the quarter of the separating employer's mass layoff). Workers who separated, but not in connection with a mass layoff, are dropped from the displaced worker "treatment" group because, for these workers, the decision to separate is more likely to have been the result either of worker choice or employer selection.

An employer is counted as having a mass layoff in a quarter in which two conditions are met (Davis and von Wachter 2012). First, during 2008–2010, employment must have dropped by 30 percent or more compared with the quarter of 2007 in which employment was greatest. Second, the employer's maximum employment in 2007 must have been less than 130 percent of that employer's maximum employment in 2006. The second condition helps to avoid classifying employers in steady decline as experiencing a mass layoff.

Because this definition of mass layoff hinges on percentage changes in the *level* of employment, small employers may be counted as having a mass layoff with only a small absolute change in employment. Accordingly, we drop from the analysis any worker who at any time had a primary employer whose employment dropped below 50 workers in any quarter during 2002–2007.

For all quarters starting with 2008:I, we require displaced workers to have at least one quarter per calendar year with positive earnings or hours to remain in the sample. This again follows JLS and implies that the estimates should be interpreted as effects of displacement on

workers who remain attached to the Washington labor force.<sup>13</sup> (In section 4.5, we estimate the effects of displacement without imposing this requirement.)

The comparison group consists of long-tenure workers (at least six years of tenure with the same primary employer between 2002–2007) who were not displaced and who continued to have positive earnings with the same primary employer in every quarter from 2008:I through 2014:IV. The comparison, then, is between the outcomes of long-tenure displaced workers and long-tenure non-displaced workers who retain employment with the same primary employer for another seven years.<sup>14 15</sup>

About one-third of the workers who we count as displaced claimed UI benefits and registered with the public labor exchange at some time during 2002–2014. We focus on this UI claimant sample because (as described above) we observe these workers' age, race, sex, and education. This allows us to perform subgroup analyses, as well as to make our analysis similar to previous research by restricting attention to displaced workers aged 20–50 at the time of displacement (and non-displaced workers aged 20–50 in 2007:IV).<sup>16</sup>

<sup>&</sup>lt;sup>13</sup> Workers who drop out of the labor force, become self-employed, work in the underground economy, or move out of state will not appear in the Washington wage records. (Self-employed workers are not covered by UI, underground earnings are not reported, and out-of-state earnings will be picked up in the wage records of another state.)

<sup>&</sup>lt;sup>14</sup> We have also developed estimates using a broader comparison group that need not remain with the same primary employer from 2008:I through 2014:IV—see Appendix C. Specifically, the comparison group includes long-tenure workers who separated (but not in connection with a mass layoff) during 2008:I through 2010:IV or who separated *for any reason* after 2010:IV. Because this comparison group is defined without conditioning on continuous future positive earnings or hours, estimated earnings losses are somewhat smaller than those estimated using the first comparison group. Krolikowski (2018) discusses the choice of a comparison group for displaced workers in greater detail.

<sup>&</sup>lt;sup>15</sup> We have conducted a robustness check that excludes the non-displaced co-workers of displaced workers from the comparison group. This exclusion drops about 20 percent of the original comparison group, and produces slightly larger estimates of displaced workers' earnings, hours, and wage rate losses.

<sup>&</sup>lt;sup>16</sup> Estimates using a broader (and considerably larger) sample—that is, not restricted to UI claimants—are similar to those obtained using the UI claimant sample. See Appendix Figures A.4, A.5, and A.6.

#### 2.2 Summary statistics for displaced workers and the comparison group

Table 1 displays descriptive statistics of key variables for the full UI claimant sample (columns 1 and 2) and for two subsamples we examine later for reasons that will become clear: the UI claimant sample excluding workers in NAICS industries 51–56 (information, finance and insurance, real estate, professional, scientific, and technical services, management of companies; administrative, support, and waste management and remediation services—columns 3 and 4); and workers whose employers paid top-quintile earnings premiums (see section 3.2—columns 5 and 6). The full UI claimant sample includes 3,032 displaced workers and 13,290 non-displaced workers. Note that all workers in the non-displaced comparison group claimed UI at some time during our observation window, as did the displaced workers. Inspection of the data shows that the non-displaced workers who claimed UI experienced relatively short spells of unemployment before being recalled to the employer from whom they were laid off.<sup>17</sup>

The top panel of Table 1 shows sample means and standard deviations of workers' quarterly average earnings, work hours, and hourly wage rate for the pre-displacement years 2002–2005. (We omit 2006–2007 to avoid including lower earnings and hours that may occur due to pre-displacement "Ashenfelter's dips.") Displaced workers had somewhat higher earnings and work hours, and substantially higher hourly wage rates, than did non-displaced workers. These differences nearly disappear, however, when workers in NAICS industries 51–56 are dropped from the sample (columns 3 and 4).

The demographic characteristics of the UI claimant sample fit the well-known profile of displaced workers: 70 percent male, 77 percent white, 45 percent with a high school education but no post-secondary education, 11 percent with less than high school or a GED, and averaging

<sup>&</sup>lt;sup>17</sup> The median UI claim duration of non-displaced workers in the sample was 2 weeks, whereas median UI claim duration of displaced workers in the sample was 4 weeks.

40 years of age.

The bottom panel of Table 1 shows two substantial differences between the displaced worker sample and the non-displaced comparison group. First, the employers of displaced workers were smaller on average than those of non-displaced workers. This difference arises because, as noted earlier, small employers are more likely than large employers to satisfy the definition of a mass layoff. Second, the distribution of displaced and non-displaced workers differs by major industry of employment in 2007:IV. About 83 percent of displaced workers came from just three major industries: NAICS codes 31-33 (manufacturing; 27 percent), 42-49 (trade; 15 percent), and 51–56 (information, finance and insurance, real estate, professional services, management, and administrative support; 41 percent). In contrast, only two-thirds of the non-displaced comparison workers worked in these industries. The imbalance results mainly from NAICS industries 51–56, which employed 41 percent of the displaced workers, but only 6 percent of the non-displaced comparison group. The relative dearth of stably employed workers in NAICS industries 51–56 who are available for comparison with workers displaced from NAICS industries 51–56 suggests that employment in those industries is less durable than employment in industries, like manufacturing, from which workers traditional have been displaced.

The composition of displaced workers in the Washington samples differs sharply from the composition of the Pennsylvania workers examined by JLS, 75 percent of whom came from manufacturing; however, in the Connecticut sample analyzed by CP is more like the Washington sample: 16 percent from manufacturing, 19 percent trade, and 23 percent from NAICS codes 51– 56. As a check on the overall estimates, we conduct an analysis of workers displaced from all industries except NAICS 51–56 in section 4.3.

#### **3** Estimation methods

This section begins with a description of methods used to estimate the effects of displacement on earnings. We next discuss the decomposition of earnings losses into components due to lower work hours and hourly wage rates. We then describe our use of AKM methods to estimate the importance of employer effects in displaced workers' employment outcomes.

#### 3.1 Estimated displacement effects on earnings, hours, and wage rates

To estimate displaced workers' earnings losses, we apply the multi-period difference-indifferences estimator from JLS, which compares the employment outcomes of displaced workers before, during, and after displacement with observationally similar long-tenure workers who were not displaced. For the UI claimant subsample, the effect of displacement can be obtained by estimating a worker fixed-effects model of the following form:

$$Y_{ijt} = c_i + \gamma_t + \mathbf{Z}_{it} \boldsymbol{\theta}_1 + \mathbf{W}_{it} \boldsymbol{\theta}_2 + \mathbf{X}_{j(i,t)} \boldsymbol{\beta} + \sum_{k=-20}^{20} (\delta_k \cdot D_{itk}) + e_{ijt}$$
(1)

where *Y* is an employment outcome (earnings, hours, or wage rate) of worker *i* (with primary employer *j*) in quarter *t*; *c<sub>i</sub>* is a worker-specific fixed effect;  $\gamma_t$  is a vector of calendar quarter indicators;  $\mathbf{Z}_{it}$  includes the worker's age and age squared, and a vector of gender, race, and education indicators, interacted with the worker's age;  $\mathbf{W}_{it}$  includes averages of the worker's predisplacement (2002–05) earnings and pre-displacement hours with the primary employer, both interacted with a vector of yearly indicators; and  $\mathbf{X}_{j(i,t)}$  consists of the characteristics of worker *i*'s pre-layoff employer *j* (logarithm of employer size and one-digit NAICS code in 2007:IV interacted with a vector of yearly indicators). Each  $D_{itk}$  is an indicator equal to one if the worker is observed in quarter *k* relative to displacement, zero otherwise (*k* = 0 is the quarter of

displacement).<sup>18</sup> (The estimating equation for the full sample is the same as equation (1) except that we necessarily drop worker characteristics  $Z_{it}$ .) Because we cannot assume the regression errors  $e_{it}$  are independent over time, we cluster standard errors at the individual worker level.

Interest lies mainly in the estimates of  $\delta_k$ , which are regression-adjusted differences in outcomes between displaced and non-displaced workers before (k < 0), at the time of (k = 0), and after (k > 0) the quarter of displacement. Interpreting the estimated  $\delta_k$  as causal effects of displacement requires the assumption that, absent displacement, the time path of displaced workers' employment outcomes would have remained parallel to the time path of non-displaced workers' outcomes. By differencing away any pre-displacement gap between displaced and stably employed workers, we can interpret post-displacement differences in outcomes as the effect of displacement. In practice, if we observe negative estimated  $\delta_k$ s after displacement, but not in the years preceding displacement, we take this as evidence of a displacement effect.<sup>19</sup>

Figure 1 illustrates the parallel-trends assumption using earnings (unconditional on employment, so including zero earnings) and hours data for workers displaced in 2009:I and workers who remain stably employed by the same employer. During the first 5-6 years of the seven before displacement, the earnings and hours of workers who will be displaced and those who will remain stably employed are roughly parallel. The profiles of both earnings and hours give a first impression that, following displacement, workers' earnings and hours worked tend not to recover to their pre-displacement levels.

<sup>&</sup>lt;sup>18</sup> The omitted reference category consists of non-displaced workers and all observations recorded in guarters 21, 22, 23, and 24 before the displacement (k < -20); hence, we limit the analysis sample to observations recorded between -24 and 20 quarters relative to the displacement quarter.

<sup>&</sup>lt;sup>19</sup> As a robustness check of the parallel-trends assumption, we also estimate a version of the worker fixed-effects difference-in-differences model with worker-specific trends (a "random trend" model):

 $Y_{it} = c_i + \omega_i t + \gamma_t + \sum_{k=-20}^{20} (\delta_k \cdot D_{itk}) + e_{it}, \qquad (2)$ where t is a (quarterly) time trend,  $\omega_i$  is a worker-specific annual growth rate over the period, and  $\gamma_t$  is a vector of calendar quarter indicators. [Other notation is the same as for equation (1).] In practice, we de-mean all variables in equation (2), then apply a fixed-effects estimator to the de-meaned data; see JLS (1993a, p. 694), Wooldridge (2002, pp. 315–317). Results are shown in Appendix Figures A.1, A.2, and A.3.

Because earnings are the product of hours worked and the hourly wage rate, the time paths of earnings and work hours imply a time path of the hourly wage rate. Accordingly, we plot the estimated time paths of earnings and of work hours for each phase of the displacement process, then obtain the time path of wage rates as a residual. We also estimate the time path of hourly wage rates directly using models like equation (1) for the level and log of hourly wage rates, recognizing that these estimates are conditional on employment.

#### 3.2 Employer fixed effects

A growing body of research has examined the importance of employers in earnings determination (e.g., Abowd, Kramarz, and Margolis 1999; Card, Heining, and Kline 2013; Card, Cardoso, and Kline 2016; Sorkin 2016; Barth et al. 2016). In general, this line of research has shown that "where you work" is important to "what you earn." The Washington data allow us to construct a linked employer-employee panel and estimate AKM models of earnings, hours, and hourly wages using data for 2002–2014 (see Appendix B for details). The resulting estimated employer fixed effects allow us in turn to observe the extent to which the earnings, hours, and wage-rate losses of displaced workers result from working for post-displacement employers with policies regarding earnings, hours, and wage rates that differ systematically from the pre-displacement employer.

The AKM models we estimate can be written:

$$\log Y_{ijt} = \alpha_i + \psi_{j(i,t)} + \theta_t + u_{ijt} , \qquad (3)$$

where *Y* is an employment outcome (earnings, hours, or the wage rate) of worker *i* (with primary employer *j*) in year *t*;  $\alpha_i$  is a worker-specific fixed effect (reflecting the productive characteristics of the worker that can be transferred between employers);  $\psi_{j(i,t)}$  is an employer-specific fixed

effect (reflecting employer characteristics that result in above- or below-average earnings, hours, or wage rates for all workers at employer *j*);  $\theta_t$  is a vector of calendar year indicators; and  $u_{it}$  is the error component. The function *j*(*i*,*t*) indexes the employer effect for worker *i* in year *t*.<sup>20</sup>

Estimation of equation (3) for each of the three outcomes results in three vectors of estimated employer fixed effects  $(\widehat{\psi}_j)$  one each for the log of earnings, log of hours, and log of wage rates (all necessarily conditional on employment). The estimated employer fixed effects for earnings or wage rates can be interpreted as measures of the advantages—premiums or rents— derived from being employed by a given employer. The estimated employer fixed effects for hours represent employers' differing policies regarding work hours. The estimates of employer fixed effects for earnings, hours worked, and wage rates are the parameters of main interest in equation (3).<sup>21</sup>

To estimate equation (3), we apply Card, Heining, and Kline's (2013) algorithm for estimating the AKM model, which is based on Abowd, Creecy, and Kramarz (2002).<sup>22</sup> The employer effects in equation (3) are identified only within a "connected set" of employers—that is, employers linked by worker transitions. We restrict the analysis to the largest connected set, which consists of 64 percent of all employers and about 90 percent of all workers in Washington. [These percentages are similar to those obtained by Sorkin (2016) using the U.S. LEHD data for

<sup>&</sup>lt;sup>20</sup> To reduce the computational burden, we estimate equation (3) using annualized data; see Appendix B. Note that because we do not have demographic information for most workers in dataset used for the AKM analysis, equation (3) includes no demographic controls. We show in Appendix B that the variance decomposition of earnings in the Washington data (without demographic controls) is very similar to that obtain by Sorkin (2016) using U.S. LEHD data for 2000–08.

<sup>&</sup>lt;sup>21</sup> See Goldschmidt and Schmieder (2017), who take a similar approach to estimate the loss of employer-specific effects due to outsourcing of jobs, and Fackler et al. (2017), who study employer-specific effects for workers displaced in connection to employer bankruptcy.

<sup>&</sup>lt;sup>22</sup> The Card, Heining, and Kline algorithm is coded in MatLab and uses tools from graph theory to find the largest connected set, then estimates the fixed effects by solving the normal equations numerically (incomplete Cholesky factorization with preconditioned conjugate gradient, which is good for situations where the matrix is large but sparse). We have used an adapted version of the MatLab code available on Patrick Kline's website: <a href="http://eml.berkeley.edu//~pkline/papers/code\_CHK.zip">http://eml.berkeley.edu//~pkline/papers/code\_CHK.zip</a> (last accessed: December 7, 2016).

2000–08.] To avoid creating a mechanical relationship between earnings changes and estimated  $\psi_j$ s, we exclude displaced workers and the non-displaced comparison group when estimating the AKM model.

We treat the estimated employer fixed effects,  $\hat{\psi}_j$ , as additional outcomes of the displacement process: the question is, what proportion of the earnings, hours, and wage rate losses following displacement can be attributed to a displaced worker's reemployment by an employer with a different  $\hat{\psi}_j$  (for earnings, hours, or wage rates) than the employer from which she was displaced. We begin by assigning the appropriate  $\hat{\psi}_j$ s (for employer *j*) to each worker-quarter observation in the displaced worker data. It is possible to assign  $\hat{\psi}_j$ s to all worker-quarter observations in the pre-displacement period (2002–2007) and to all but 622 worker-quarter observations in the post-displacement period. (All 622 observations are for displaced workers in the post-displacement period.) The unmatched cases occur when the employer of a displaced worker was not in the connected set used to estimate the AKM  $\hat{\psi}_j$ s. There are 124 such employers, and they employed altogether 160 unique displaced workers for at least one quarter following displacement.<sup>23</sup>

To estimate the importance of employer-specific fixed effects in explaining the adverse outcomes of displaced workers, we compare the employer fixed effects (for earnings, hours, and

<sup>&</sup>lt;sup>23</sup> When we estimate the AKM model using data only from the pre-displacement years (2002–2007), rather than from all available years (2002–2014), we are unable to match 22,806 worker-quarter observations for 1,762 unique displaced workers who were employed by 834 employers in the post-displacement period. For these 834 employers,  $\hat{\psi}_j$ s could not be produced either because they were not in the connected set or because they did not exist before 2008. (As with the 2002–2014 data, 2002–2007 data produces  $\hat{\psi}_j$ s for all pre-displacement employers.) The correlation coefficients between  $\hat{\psi}_j$ s estimated using 2002–2014 data and  $\hat{\psi}_j$ s estimated using 2002–2007 data are 0.98 for log earnings, 0.96 for log hours, and 0.96 for log wage rates.

wage rates) of displaced workers after displacement with the employer fixed effects beforedisplacement (and relative to the employer fixed effect of workers not displaced). That is, for every quarter k relative to displacement, we compute:

$$\mathbf{E}\left[\psi_{j} \mid \alpha_{i}, \theta_{t}, D_{itk} = 1\right] - \mathbf{E}\left[\psi_{j} \mid \alpha_{i}, \theta_{t}, D_{itk} = 0\right],$$

where  $D_{itk}$  equals one if the observation is for a displaced worker observed in a postdisplacement quarter, and zero if the observation is for a stably employed worker or a displaced worker observed in a pre-displacement quarter. In practice, we regress the estimated employer fixed effect  $\hat{\psi}$  on pre- and post-displacement indicators, along the lines of equation (1):

$$\widehat{\psi}_{ijt} = c_i + \gamma_t + \mathbf{Z}_{it} \mathbf{\theta}_1 + \mathbf{W}_{it} \mathbf{\theta}_2 + \mathbf{X}_{j(i,t)} \mathbf{\beta} + \sum_{k=-20}^{20} (\delta_k \cdot D_{itk}) + e_{ijt}$$
(4)

The estimated  $\delta_k$ s are regression-adjusted differences in employer fixed effects realized by displaced workers relative to non-displaced workers and relative to before displacement. Equation (4) includes individual worker fixed effects, so the estimated  $\delta_k$ s represent within-worker changes in employer fixed effects (for earnings, hours, and wage rates) resulting from transitions from pre- to post-displacement employers. For earnings and wage rates, we interpret negative estimated  $\delta_k$ s as evidence of lost employer-specific premiums (which could result from lost rents or compensating differentials; see Sorkin 2016). For hours worked, we interpret negative estimated  $\delta_k$ s as evidence of reduced hours due to the working time policies of post-displacement employers.

#### 4 Estimated effects of displacement on earnings, work hours, and wage rates

This section presents estimates of the magnitude of displaced workers' earnings losses and decomposes those losses into their work-hour and hourly wage rate components. We also examine the effects of displacement on "traditional" displaced workers (that is, workers displaced from industries other than NAICS industries 51–56), workers with relatively short predisplacement job tenure, and workers with relatively weak labor force attachment. In section 5, we examine the extent to which displaced workers' earnings losses can be attributed to employer fixed effects.

#### 4.1 Estimates of lost earnings

Figure 2 displays estimated effects of displacement on unconditional earnings (top) and log earnings (bottom) over a period of 5 years, and Table 2 offers a summary of the estimates for the quarter following displacement (quarter 1) and the average of quarters 17–20 following displacement. The graphs are obtained by estimating equation (1) and plotting the estimated  $\delta_k$ s, along with 95-percent confidence intervals (which are very small and at times hard to see). The vertical line in each graph marks the quarter of displacement; that is, the last quarter in which a displaced worker is observed with earnings or hours with the employer of the previous six years.

A clear pattern emerges from Figure 2. Soon-to-be-displaced workers' earnings dip (or drift downward slightly) in roughly the year before displacement (Ashenfelter's dip), drop sharply in the quarter of displacement and the quarter immediately following displacement (quarters 0 and 1), then recover, but never to their pre-displacement level, as gauged relative to earnings of the comparison group. It follows that the earnings losses of displaced workers have three sources: Ashenfelter's dip before displacement (which is small in this case), the drop at and

immediately after separation, and most significantly, the long-term earnings deficit after reemployment (see again Figure 1).

The top graph in Figure 2 shows that, in the quarter following displacement, workers lost on average about \$5,960 compared with comparable non-displaced workers. (See also the top panel of Table 2 and Appendix Table A.1.) Dividing this by pre-displacement (2002–05) average earnings with the former primary employer (\$12,482, from Table 1) implies a loss of about 48 percent in the quarter following displacement. Estimates for the log of earnings are somewhat smaller than the other estimates, suggesting that, in the quarter after displacement, earnings are on average less than those of the comparison group by about 39 percent [exp(-0.488) - 1]; however, because these estimates condition on positive earnings, they understate the extent of earnings losses.

The above estimate of earnings losses in quarter 1 is based solely on earnings from the primary employer in the quarter following displacement. (This employer by definition cannot be the employer from which the worker was displaced.) If we instead examine earnings from all post-displacement employers, estimated earnings losses are similar, although somewhat smaller (see again Appendix Table A.1). This suggests that accounting for the presence of multiple employers does not substantially change the conclusions drawn from focusing solely on outcomes from primary employers.

Figure 2 and Table 2 also shows that, five years after displacement, displaced workers earned on average \$1,940 less per quarter from their primary employer than did comparable nondisplaced workers. This translates to lost earnings of about 16 percent (dividing by \$12,482). The

log earnings estimates suggest long-term losses of about 15 log points.<sup>24</sup> As was true for earnings in the quarter after displacement, estimated long-term earnings losses based on earnings from all employers are very similar to those based on earnings losses from only the primary employer (Appendix Table A.1).

Table 3 compares the estimates in Figure 2 and Table 2 with those obtained by JLS and Couch and Placzek (2010, hereafter "CP"), the studies most similar to ours. JLS (1993b, Figure 5.5) report that, at the time of displacement, their sample of Pennsylvania UI claimants experienced a drop in earnings of about \$7,800 (in 2010 dollars, \$4,000 in 1987 dollars), and five years after the displacement, their earnings remained about \$2,900 (in 2010 dollars, \$1,500 in 1987 dollars) lower than they would have been otherwise. CP (Figure 4) show that, immediately following displacement, the quarterly earnings of their sample of Connecticut UI claimants were about \$7,700 (in 2010 dollars; \$6,000 in 2000 dollars) less than they would have been otherwise—a 49 percent deficit. Five years after displacement, these workers still earned about \$5,100 (in 2010 dollars; \$4,000 in 2000 dollars) less than otherwise—a 32 percent deficit.

The earnings losses we estimate for Washington workers displaced in 2008–2010 are somewhat smaller than those reported by either JLS or CP, but the similarities are perhaps more striking than the differences.<sup>25</sup>

<sup>&</sup>lt;sup>24</sup> Appendix Figures A.1–A.3 repeat the analysis in Figures 2–4 using the random trend model; see equation (2). The estimated time paths are similar to those estimated using equation (1). This similarity suggests that, before displacement—and conditional on characteristics of the worker and pre-layoff employer ( $Z_{ii}$  and  $X_{iji}$ )—earnings and work hours of displaced and non-displaced workers evolve approximately in parallel. This tends to buttress the interpretation of post-displacement outcome differences between displaced and non-displaced workers estimated using model (1) as the effects of displacement.

<sup>&</sup>lt;sup>25</sup> The JLS sample consists largely of workers displaced from manufacturing in Pennsylvania during the time of the decline of the U.S. steel industry, and the CP sample, although more diverse, consists disproportionately of workers displaced from shipbuilding during the 2001 recession. No single industry in Washington imploded during the Great Recession, although Washington clearly experienced a severe contraction.

#### 4.2 Estimates of lost work hours and reduced hourly wage rates

Figure 3 displays estimated effects of displacement on unconditional work hours (top) and log hours (bottom), again based on equation (1). (See also the top panel of Table 2. Appendix Table A.2 shows the estimates in detail.) As was true for earnings, the work hours of soon-to-be-displaced workers dip somewhat in roughly the year before displacement, drop greatly in the quarter of displacement and the quarter following displacement, then partially recover. The recovery of work hours appears more robust than the recovery of earnings, but work hours of the displaced workers nevertheless remain below those of the non-displaced comparison group.

A comparison of Figures 2 and 3 suggests that most of the earnings losses in the quarter of displacement and the quarter following can be explained by lost work hours. Lost hours in the quarter after displacement amount to 200 hours on a base of 500 hours (40 percent, estimated in levels), or about 38 log points. The corresponding earnings losses are about 48 percent (in levels), or about 49 log points, so we infer that lost work hours are responsible for roughly 80 percent of lost earnings around the time of displacement.

Five years after displacement, reduced work hours remain an important reason for lost earnings. Figure 3 shows that after five years (in quarters 17–20), the average displaced worker still works 29 fewer hours per quarter than otherwise (about 6 percent, estimated in levels), or about 7 log points. The corresponding earnings losses are about 16 percent (in levels), or about 15 log points, so we infer that reduced work hours account for about 45 percent (7 out of 15 log points) of the long-term earnings losses of displaced workers. It follows that the remaining 55 percent of long-term earnings losses are accounted for by lower hourly wage rates (see below).

These estimates may be surprising in light of Stevens's (1997) finding that reduced work hours (as opposed to declining wage rates) play a relatively minor role in explaining displaced

workers' long-term earnings losses. However, Farber (2017) also finds that reduced hours (conditional on working) are a factor in explaining earnings losses—specifically, he finds that the movement from full-time to part-time employment explains a significant part of the cost of displacement during and after the Great Recession.<sup>26</sup>

To further investigate displacement's effects on work hours, we estimate a set of linear probability models, based on equation (1), using indicators of whether a worker's weekly work hours (averaged over a quarter) exceeded a given threshold. In particular, we estimate models for weekly hours exceeding, 0, 10, 20, 30, 35, 40, 45, 50, and 60. The estimates are displayed in Table 4 and suggest three main points about the way displacement changes the distribution of hours in the long-term (quarters 17–20). First, the overall long-term loss of work hours appears not to be driven by non-employment (i.e., more workers with 0 hours): although displacement reduces the probability of any work (Hours > 0) by about 2 percentage points in quarters 17–20, this is a small reduction in proportional terms (0.02 on a base of 0.996). Second, the main effect of displacement is to reduce the probability of working at least 20, 30, 35, and 40 hours per week. This is surprising because displacement implies lost seniority, which would imply loss of overtime hours, but it appears not to be the case. In fact, the estimates for hours in excess of 50 and 60 suggest that displacement may slightly *increase* the probability of work hours at the high end of the hours distribution; that is, if anything displaced workers are slightly more likely to work overtime after displacement. This is the third main implication of the estimates in Table 4.

<sup>&</sup>lt;sup>26</sup> Farber's (1993, 2015, 2017) studies makes use of the Displaced Worker Supplement to the Current Population Survey, so they pertain to are broader group of workers—those who previously lost jobs due slack work, plant closing, or elimination of a position or shift—than the studies using UI administrative data, which are restricted to long-tenure workers who separate in connection with the employer experiencing a large contraction. For this and other reasons, comparisons between Farber's estimates and those presented here are not straightforward.

The conclusion above—that displaced workers' reduced hours explain about 45 percent of their long-term (five-year) earnings deficit—implies that lower hourly wage rates should explain the remaining 55 percent of the long-term earnings deficit. As a check, Figure 4 shows estimated hourly wage rate losses from equation (1) estimated using the log of the hourly wage rate as the dependent variable. The results suggest that, five years after displacement, hourly wage rates are about 8 log points lower than otherwise (see the top panel of Table 2). These estimates are best compared with those using log earnings and log hours as outcomes (since they are also conditioned on employment), and together they again imply that about 45 percent of the long-term earnings deficit of 15 percent is due to fewer work hours, and about 55 percent to lower hourly wage rates.<sup>27</sup> Together, the long-term time paths of log hours worked and log hourly wage rates imply a high labor supply elasticity. We return to this point in section 4.4.

The estimated hourly wage profile (Figure 4) differs from both the earnings profile in Figure 2 and the work hours profile in Figure 3. Following displacement, log hourly wage rates drop immediately by about 10 percent and remain permanently lower by about 7 percent. This differs from the earnings and hours profiles, which first drop, then partially recover after two years. We explore whether employer fixed effects are the source of this persistent long-term drop further in Section 5.

<sup>&</sup>lt;sup>27</sup> Figure 4 also shows a clear 6 log-point spike in hourly wage rates in the quarter of displacement (quarter 0). This spike results from a greater drop in work hours than in earnings in the quarter of displacement: for example, Figure 2 shows an earnings drop of about 36 log points in quarter 0, whereas in Figure 3 shows a work-hours drop of about 40 log points. Severance payments paid in the quarter of separation are the most likely cause of this pay bump. Severance payments are included with earnings in administrative wage records and would inflate reported earnings relative to work hours in the quarter of separation. The result is an apparent increase in the hourly wage around the time of displacement.

#### 4.3 Estimates excluding workers displaced from NAICS industries 51–56

To make our estimates more comparable with those of JLS and CP, we repeat the analysis *excluding* workers displaced from jobs in NAICS industries 51–56 (information, finance and insurance, real estate, professional, scientific, and technical services, management of companies; administrative, support, and waste management and remediation services). We do this for two reasons: first, as was seen in Table 1, workers in NAICS industries 51–56 have substantially higher earnings and wage rates than other workers; second, the comparison sample for workers displaced from NAICS industries 51–56 is quite thin, making inferences about the influence of displacement on these workers less convincing than for others.

Figure 5 plots the losses of these "traditional" displaced workers (that is, those displaced from industries other than NAICS 51–56—see also the second panel of Table 2). Figure 5 suggests these workers experience immediate earnings losses of nearly 60 log points, and long-term earnings losses (quarters 17–20) of more than 20 log points. For workers displaced from industries other than NAICS 51–56, then, the long-term losses appear larger than for the overall sample. Still, these estimated long-term losses are somewhat smaller than for the Pennsylvania and Connecticut workers studied by JLS and CP.

Figure 5 shows that the long-term lost work hours and reduced wage rates of traditional displaced workers also exceed those for workers overall. Their long-term hours loss is about 10 log points (compared with 7 log points for all workers), and their long-term wage reduction is about 11 log points (compared with 8 log points for all workers).

#### 4.4 Losses of workers with relatively short job tenure

Displaced workers' long-term losses have been attributed to lost employer-specific and industryspecific human capital (e.g., Topel 1990; Neal 1995; Carrington and Fallick 2017). The implication of the specific human capital hypothesis is that longer pre-displacement tenure is associated with a larger loss of earnings. Farber (1993) found that, on average, each additional year of pre-displacement job tenure was associated with an additional one percent drop in postdisplacement earnings. In contrast, in a study using administrative data, von Wachter, Song, and Manchester (2009) find insubstantial differences between the earnings losses of workers with three years of tenure and workers with six or more years of tenure.

We next examine and compare the earnings, hours, and wage rate losses of workers with 6 or more years of pre-displacement job tenure (the main sample) with the losses of workers with only 3–4 years of pre-displacement tenure. To do this, we estimate equation (1) separately for each of the two displaced worker groups (long-tenure and shorter-tenure), using the same non-displaced workers with 6 or more years of tenure as the comparison group. We view this as a descriptive exercise rather than an attempt to estimate the effect of job tenure on the outcomes of displaced workers.

Figure 6 shows the estimated profiles of earnings, work hours, and wage rates (in logs), and the estimated time paths of the quarterly employment probability. The earnings losses and employment probabilities of displaced workers with 3–4 years of tenure are quite similar to those with 6 or more years of tenure, consistent with the findings in von Wachter, Song, and Manchester (2009). However, the patterns of hours losses and wage rate reductions differs sharply between the two groups. The hours losses of short-tenure displaced workers are consistently greater than the hours losses of long-tenure displaced workers. In contrast, the wage

rates of short-tenure displaced workers return to their pre-displacement level within about four years of displacement, whereas the wage rates of long-tenure displaced workers never recover. Further, Figure 6 shows that hourly wage rates of workers with 3–4 years of tenure recover five years after displacement. This differs from the pattern of wage losses experienced by workers with longer tenure and is not explained by a difference in employment probabilities.

One possible interpretation of these estimates is that the reduced work hours of longtenure displaced workers represent a labor supply response to their reduced wage rates, whereas the substantially reduced hours of short-tenure displaced workers, along with wage rates similar to those faced before displacement, suggests demand constraints faced by these workers. An implication would be that short- and long-tenure displaced workers differ in ways that should not be attributed to previous job tenure alone.<sup>28</sup>

#### 4.5 Losses of workers less strongly attached to the labor force

The main estimates in Figures 2–4 are based on a sample of displaced workers who were employed in at least one quarter per calendar year in each year following displacement. How sensitive are the estimates to this restriction? Relaxing it, so that a displaced worker need never reappear with positive earnings or hours after being displaced, increases the sample of displaced workers from 3,032 to 4,199. Figure 7 plots the estimated time paths of log earnings, log hours, and log wage rates for the unrestricted sample in red, along with the time paths of earnings, work hours, and wage rates for the restricted sample (repeated from Figures 2–4) in blue.

<sup>&</sup>lt;sup>28</sup> We have also examined losses due to displacement separately for workers younger than age 40 in the quarter of displacement and for workers age 40 and older in the quarter of displacement. (To construct the non-displaced comparison groups, we use age in 2007:IV.) The estimated long-term earnings, hours, and wage-rate losses of the younger and older workers are quite similar, which is surprising because older workers have on average longer job tenure. However, the reemployment rates of older workers in the first two years after displacement are lower than those of younger workers, consistent with Farber's (2017) findings.

The most striking difference between the restricted and unrestricted samples is that, five years after displacement, the probability of employment—defined as having positive earnings or hours in that quarter—for the unrestricted sample is 18 percent less than for the comparison group, compared with 4 percent less for the restricted sample. But differences between the two groups in the time paths of log earnings, log hours, and log wage rates are less striking. Specifically, five years after displacement, earnings are 24 log points lower in the unrestricted sample (compared with 16 log points lower in the restricted sample)<sup>29</sup>, hours are 11 log points lower (compared with 7 log points), and wage rates are 13 log points lower (compared with 8 log points).

<sup>&</sup>lt;sup>29</sup> When Couch and Placzek (2010, p. 579) relax the labor force attachment restriction, they find earnings losses that are greater by 15–18 percentage points, substantially larger than the 8 log-point increase we estimate.

#### 5 Employer fixed effects

Section 3.2 described an approach to estimating whether some portion of displaced workers' losses are due to employer fixed effects. For earnings or wage rates, this would imply loss of a job with an employer offering premium earnings or wage rates, and reemployment with an employer that did not. For hours worked, this would imply loss of a job with systematically longer hours than the job obtained after displacement. This section describes the results of that approach. We first examine whether employer fixed effects are responsible for part of the average displaced workers' earnings losses, lost work hours, and lower hourly wage rates. Second, we examine differences in the importance of employer fixed effects for workers displaced from employers paying high and low earnings premiums. Third, we examine the losses of displaced workers who move to employers. Finally, we study the role of industry fixed effects in explaining the losses of displaced workers.

#### 5.1 Estimated losses due to employer fixed effects for the average displaced worker

The blue-dot time paths in the three panels of Figure 8 display estimated effects of displacement on the employer fixed effects ( $\hat{\psi}_j$ ) for log earnings, log hours, and log of the hourly wage rate for the average displaced worker in the sample. [These are the  $\hat{\delta}_k$ s from equation (4), estimated over the sample of all displaced workers who claimed UI.] For comparison with the main results in Figures 2–4, the red-circle time paths in Figure 8 repeat the estimated effects of displacement on log earnings, log hours, and log of the hourly wage rate with the primary employer. [The estimates underlying Figure 8 are reported in Table 5. For each post-displacement quarter, the estimates in the odd-numbered columns give the total log-point changes in earnings (column 1),

hours (column 3), and hourly wage rate (column 5) attributable to displacement, and the evennumbered columns give the log-point changes in earnings (column 2), hours (column 4), and hourly wage rate (column 6) attributable to loss of an employer offering premium earnings.]

For each of the three outcomes, lost employer fixed effects explain a modest (and statistically significant) fraction of the losses due to displacement—see the summary in row B of Table 2. In the quarter following displacement (quarter 1), about 3 log points of the overall earnings loss of 49 log points (or 6.6 percent) are due to working for an employer that pays less to all its workers, controlling for worker fixed effects. For work hours, differing employer hours policies account for about 1.5 of the overall 38 log-point loss (about 4 percent), and lost wage rate premiums account for 1.7 of the overall 11 log-point reduction (about 15 percent).

Twenty quarters after displacement, employer fixed effects play a more important role in the lost earnings and wage rates of the average displaced worker, but essentially no role with respect to lost work hours. Specifically, employer fixed effects account for 1.7 of the overall 15 log-point earnings loss (or 11 percent), and 2 of the overall 8 log-point reduction in hourly wages rates (25 percent). Employer hours policies play no direct role in the loss of work hours. Overall, the estimates suggest that employer fixed effects, although not the main reason for the average displaced workers' long-term losses, nevertheless play some role in those losses.

At the same time, the estimates displayed in Figure 8 suggest an important role for the AKM residual  $u_{it}$  (in equation (3)) in explaining the average displaced worker's losses. Because both equations (1) and (4) control for worker fixed effects, the large remaining gap between the overall losses and losses due to employer fixed effects (the gap between the red-circle and bluedot time paths in Figure 8) must be due to factors other than worker and employer fixed effects.

This in turn suggests a role for time-varying factors, such as specific worker-employer matches and lost specific human capital, in explaining displaced workers' long-term losses.

#### 5.2 Estimated losses by quintile of pre-displacement employer earnings premium

We consider next whether workers displaced from employers offering earnings premiums in the top-quintile experience greater long-term losses than other displaced workers. We also consider whether a greater share of the losses of these displaced workers are due to the loss of employer-specific premiums.

Figure 9 displays the estimated effects of being displaced from an employer whose earnings fixed effect (or premium,  $\hat{\psi}$ ) was in the top quintile of employers before 2008. (Descriptive statistics for the sample used to obtain these estimates are displayed in columns 5 and 6 of Table 1. The estimates in all three panels of Figure 9 are based on the sample of workers with a pre-2008 employer with top-quintile *earnings* premiums.) Figure 10 displays estimated effects of being displaced from an employer whose earnings fixed effect was in a lower quintile (1 through 4) of employers pre-2008. (The estimates in all three panels of Figure 10 are based on the sample of workers with a pre-2008 employer with lower-quintile *earnings* premiums.)

As before, the red-circle time paths in Figures 9 and 10 show the estimated effects of displacement on log earnings, log hours, and log of the hourly wage rate (from equation (1)); the blue-dot time paths show estimated effects of displacement on employer fixed effects for log earnings, log hours, and log of the hourly wage rate (from equation (4)).

Comparison of Figures 9 and 10 shows that the overall losses of workers displaced from top-quintile employers are somewhat less than those of workers displaced from lower-quintile employers. However, all three panels of Figure 9 show that the share of losses accounted for by

foregone employer fixed effects is substantially greater for workers displaced from top-quintile employers, both at the time of displacement and in the long term. In contrast, Figure 10 shows that essentially none of the losses of workers displaced from lower-quintile employers can be explained by foregone employer fixed effects.

Table 2 (rows E and F) provides details of the estimated losses at quarters 1 and 17–20 after displacement. For workers displaced from top-quintile employers, lost employer premiums explain about 54 percent (-0.067/-0.125) of the earnings loss, 38 percent (-0.021/-0.056) of the reduced hours worked, and 83 percent (-0.046/-0.063) of the reduced wage rate 17–20 quarters after displacement. This compares with 11 percent (-0.017/-0.152) of the earnings loss, 4 percent (-0.003/-0.070) of the reduced hours worked, and 25 percent (-0.020/-0.080) of the reduced wage rate explained by lost employer premiums for workers overall. The importance of employer fixed effects to the losses experienced by workers displaced from top-quintile employers has implications that we pursue in section 6.

#### 5.3 Losses of workers moving to employers offering lower, similar, and higher premiums

It makes sense to explore further the role of employer-specific premiums by examining the time paths of earnings, hours worked, and wage rates for workers who, two years after displacement, worked for an employer whose earnings fixed effect (or premium,  $\hat{\psi}$ ) was in the same quintile as the pre-displacement employer. We also examine the outcomes of workers who moved to employers whose earnings fixed effect was in a higher or lower quintile than that of the separating employer. This analysis is strictly descriptive because it conditions on an outcome (type of employment two years after displacement). In this sense, it echoes Card, Heining, and

Kline's (2013) event-study analysis, which plots the earnings profiles of workers who move among employers paying similar or different premiums.

Figure 11 displays estimated displacement effects for three groups of workers: those reemployed by an employer whose earnings fixed effect was in the same quintile as the predisplacement employer's (blue dots), those reemployed by an employer with a lower-quintile earnings fixed effect (red circles), and those reemployed by an employer with a higher-quintile earnings fixed effect (green hollow squares). We use the *earnings* fixed effects to classify displaced workers' changes in employer quintile for all three outcomes—log earnings, log hours, and log hourly wage.

All three groups of displaced workers experience earnings losses and lower wage rates, but the losses are smallest for workers reemployed by an employer with an earnings fixed effect in the same quintile as the pre-displacement employer. Workers reemployed by an employer with a lower earnings fixed effect experience the largest losses, and those reemployed by an employer with a higher earnings fixed effect suffer losses in between those of the other two groups. Also, although work hours recover to nearly their pre-displacement level for all three groups of workers, they recover most quickly for workers reemployed by a same-quintile employer. With respect to wage rates, the long-term losses are close to zero for workers reemployed by a samequintile employer.

Why do displaced workers reemployed by a higher-quintile fixed effect employer fare worse that those reemployed by a same-quintile fixed effect employer? A reasonable inference from this pattern is that the AKM framework, with additive worker and employer fixed effects, fits displaced workers (i.e., relatively skilled workers with long tenure) less well than it fits workers as a whole. [In Appendix B, we find evidence similar to that in Card, Heining, and Kline

(2013) that the average gain in earnings for those who move to a higher fixed effect employer is symmetric to the average loss in earnings for those who move to a lower fixed effect employer. This suggest that the AKM framework is a good fit for the Washington labor market overall.] The inability of worker and employer fixed effects to more fully explain displaced workers' losses again suggests the importance of other factors, such as lost match effects or specific human capital, in explaining those losses.

#### 5.4 Industry fixed effects

An important literature from the 1980s and 1990s has examined the role of industry of employment in wage determination and motivated much of the subsequent research using linked employer-employee panels (Krueger and Summers 1986, 1988; Dickens and Katz 1986, 1987; Murphy and Topel 1986; Katz and Summers 1989; Groshen 1991; Gibbons and Katz 1992). A central conclusion of this research is that the well-known competitive labor market model does not offer an explanation of large and persistent inter-industry wage differentials, analogous to the conclusion of the AKM literature that the competitive model does not explain persistent withinfirm wage differentials.

In this section, we examine the role of the industry from which a worker is displaced, and the loss of industry-specific fixed effects, in explaining displaced workers' earnings losses. The goal is to understand whether the losses we have been attributing to employer fixed effects are in fact the result of the loss of a job in an industry in which policies for earnings, hours, or wage rates are out of the ordinary.
To proceed, we first estimate an AKM-type model, using the full Washington linked employer-employee panel, in which employer fixed effects are replaced by industry fixed effects:

$$\log Y_{ikt} = \alpha_i + \lambda_{k(i,t)} + \theta_t + u_{ikt}, \qquad (5)$$

where  $\lambda_{k(i,t)}$  denotes 102 three-digit NAICS-industry fixed effects, reflecting features of an industry that produce above- or below-average earnings, hours, or wage rates for all workers in industry *k*. (Other notation is the same as in equation (3).) The function k(i,t) indexes the industry fixed effect for worker *i* in year *t*. Estimation of equation (5) for each of the three outcomes produces three vectors of estimated industry fixed effects ( $\hat{\lambda}_k$ ). We use these as dependent variables in equation (4), which produces estimates of changes in earnings, hours, and wage rates attributable to lost industry fixed effects (the  $\delta_k$ s in equation (4)).

Figure 12 shows the results of this procedure. The red-circle time paths repeat the estimated full displacement effects on log earnings, log hours, and log wage rates (from Figures 2–4), and the blue-dot time paths display the estimated displacement losses attributable to estimated industry fixed effects for the same outcomes. The estimates suggest that lost industry fixed effects play a modest role in the wage rate losses of displaced workers during the two years following displacement (about 1 log point of the full reduction of 10–15 log points), but little role thereafter. Further, foregone industry fixed effects play almost no role in the earnings and hours losses of displaced workers, even near the time of displacement.

The estimates in Figure 12 seem to conflict with the literature cited above showing the importance of industry of employment to wage differentials, and appear at odds with findings showing the importance of lost industry-specific human capital for displaced workers' earnings losses (Kletzer 1989; Neal 1995; Parent 2000). However, the estimated  $\lambda_k$ s from equation (5) are

34

substantial and widely dispersed, supporting the existence of significant industry-specific premiums.<sup>30</sup> Further, when we estimate displacement effects separately for workers who did and did not return to the 1-digit industry from which they separated, we find that the losses of workers who returned to their pre-displacement industry are about half those of workers who moved to a different industry.<sup>31</sup>

As a result, the estimates in Figure 12 should be interpreted only in the sense that industry fixed effects play little role in displaced workers' losses, not that industry fixed effects are broadly unimportant. In showing that industry fixed effects play little role in displaced workers' losses, they suggest that industry premiums and losses that are correlated with changing industry are correctly attributed to lost industry-specific human capital, rather than to lost industry-wide rents. Also, they tend to reinforce the conclusion that employer fixed effects (and lost rents with respect to earnings and wage rates) play a significant role in displaced workers' losses.

<sup>&</sup>lt;sup>30</sup> The AKM variance decompositions show that industry fixed effects explain 7 percent of the variation in log earnings, 10 percent of the variation in log hours, and 3 percent of the variation in log wages. Positive sorting of workers to industries explains 7 percent of the variation in log earnings, virtually none of the variation in log hours, and 9 percent of the variation in log wages. (This compares with employer fixed effects explaining 21 percent of the variation in log hours, and 13 percent of the variation in log wages—see Appendix Table B.2.)

<sup>&</sup>lt;sup>31</sup> Based on employment 8 quarters after displacement, about 49 percent of the displaced workers we examine did not return to their pre-displacement industry. For these workers, the losses appear larger for wage rates than for hours, suggesting lost industry-specific human capital could be a factor in earnings losses. The implication is not strong because whether a worker returns to her pre-displacement industry is endogenous.

#### **6** Discussion and Conclusion

We have attempted to accomplish three objectives: (a) obtain estimates of the long-term earnings losses suffered by workers displaced during the Great Recession using comprehensive UI administrative data; (b) decompose those losses into portions due to fewer hours worked and to reduced hourly wage rates; and (c) estimate the extent to which changes in displaced workers' earnings, work hours, and wage rates can be attributed to fixed unobserved characteristics of the employers that lay off and reemploy those workers.

The estimates are based on quarterly administrative wage records from Washington State during 2002–2014, which are unusual in providing information on quarterly work hours, as well as quarterly earnings, over a 13-year period. Coverage of the wage record is very broad: all UIcovered employers and workers in Washington are included, and each employer's level of employment, payroll, and detailed industry are known.

We take Jacobson, LaLonde, and Sullivan's (1993a, 1993b) well-known approach to estimating the effects of displacement on earnings, and we extend that approach to a decomposition of those earnings losses into components attributable to changed work hours and reduced hourly wage rates. In addition, we estimate employer fixed effects for earnings, work hours, and wage rates using the approach of Abowd, Kramarz, and Margolis (1999), and we use these to estimate the extent to which displaced workers' losses can be attributed to the loss of employer-specific fixed effects.

The main findings are as follows. First, we find that displaced workers experience large and persistent earnings losses. In the quarter after displacement, estimated earnings losses are about 48 percent; in the long-term (that is, 17–20 quarters after displacement), displaced workers' earnings are still about 16 percent less than those of comparable non-displaced workers.

36

The similarities between these and earlier estimates are perhaps more striking than the differences. They are quite similar to Davis and von Wachter's (2012) estimates of the earnings losses from displacement during several recessions, although they are somewhat smaller than the estimates of Jacobson, LaLonde, and Sullivan (1993a, 1993b) and Couch and Placzek (2010), who were examining the effects of displacement in states that were experiencing the severe contraction of one or more specific industries.

Second, immediately after displacement, about 80 percent of earnings losses are explained by the loss of employment and fewer hours worked. But over time, the relative importance of lost hours falls and the importance of reduced wage rates rises: five years after displacement, about 45 percent of earnings losses can be attributed to reduced work hours, and 55 percent to lower wage rates. The increasing importance of reduced wages with time since displacement reflects the underlying profiles of work hours and wage rates. Whereas work hours drop at the time of displacement, then partially recover (see Figure 3), wage rates drop at the time of displacement and recover only slightly (Figure 4).

Third, for the average displaced worker, stable employer-specific characteristics (fixed effects) account for 11 percent of the long-term earnings losses and about 25 percent of the long-term lower wage rates due to displacement. But for workers displaced from high-premium employers (that is, employers paying earnings premiums in the top quintile), lost employer premiums explain about 54 percent of the long-term earnings losses, 38 percent of the reduced hours worked, and 83 percent of the reduced wage rates five years after displacement. Given that about 60 percent of the displaced workers we observe were displaced from top-quintile

37

employers, we interpret this as evidence that lost employer rents are an important part of the losses of most displaced workers.<sup>32</sup>

Two significant implications follow from these findings. First, for the average displaced worker, employer fixed effects explain only about 25 percent of long-term wage-rate losses (about 8 log points). This suggests a large role for an unobserved, time-varying job-specific factor, such as lost firm-specific human capital or loss of a favorable specific job-match, in explaining displaced workers' long-term earnings losses (see, e.g., Jung and Kuhn forthcoming). The lost specific human capital interpretation is supported by the finding that displaced workers with short job tenure do not experience long-term wage-rate losses, whereas longer-tenure displaced workers do. The lost job-match interpretation is supported by the finding that displaced workers who are reemployed by a higher fixed-effects employer, do not experience a wage rate gain, as would be implied by the additively separable AKM model; see Card, Heining, and Kline (2013) for a discussion. The data do not allow us to distinguish directly between the lost specific human capital interpretation and the lost job-match interpretation.

The second implication of the findings is that the labor supply of workers displaced in Washington during the Great Recession is quite elastic: the average displaced worker experiences an 8 percent reduction in her hourly wage rate, and a 7 percent reduction in work hours. The interpretation of these estimates as a labor supply effect is supported by the finding that employer-specific policies regarding hours worked (employer fixed effects for hours) play a minor role in explaining displaced workers' reduced work hours.

<sup>&</sup>lt;sup>32</sup> Our findings differ from Fackler et al. (2017) who find that (i) the losses of displaced workers in Germany are negligible and (ii) almost entirely explained by the loss of employer effects. However, Fackler et al. (2017) study displacement due to employer bankruptcy, a special case that complicates direct comparison. In a related paper, Fackler et al. (2016, p. 128) argue, that from the worker's perspective, mass layoffs (defined as a thirty percent drop the employer workforce for employers employing more than 50 workers) are a better proxy for unanticipated exogenous separation shocks than are employer bankruptcies.

The findings have some unusually clear implications for policy. First, for the average displaced worker, earnings losses can be attributed mainly to either lost firm-specific human capital or the loss of a favorable job match. For these workers, effective reemployment services are a logical policy approach because of the need either to reestablish a favorable job match or to become reemployed and start re-accumulating specific human capital. Second, for workers displaced from employers who paid premium earnings and wage rates, a wage-rate subsidy is a reasonable alternative because it would both compensate displaced workers for their loss and act to offset their reduced labor supply (Davidson and Woodbury 2000). Retraining is a relatively unattractive alternative, given the lack of evidence that deficient general human capital is at the root of displaced workers earnings losses. In fact, the importance of employer fixed effects in explaining the losses of the 60 percent of workers who were displaced from an employer paying a top-quintile earning premium provides a possible explanation for the limited effectiveness of retraining in improving the earnings of displaced workers.

#### References

Abowd, John M., Robert H. Creecy, and Francis Kramarz. 2002. "Computing Person and Firm Effects Using Linked Longitudinal Employer-Employee Data." U.S. Census Bureau Technical Paper on Longitudinal Employer-Household Dynamics, No. 2002-06.

Abowd, John M., Francis Kramarz, and David N. Margolis. 1999. "High Wage Workers and High Wage Firms." *Econometrica* 67(2): 251–333.

Abraham, Katharine, G., John C. Haltiwanger, Kristin Sandusky, and James Spletzer. 2016. "The Consequences of Long Term Unemployment: Evidence from Matched Employer-Employee Data." NBER Working Paper No. 22665.

Barth, Erling, Alex Bryson, James C. Davis, and Richard Freeman. 2016. "It's Where You Work: Increases in the Dispersion of Earnings across Establishments and Individuals in the United States." *Journal of Labor Economics* 34(S2): S67–S97.

Card, David, Joerg Heining, and Patrick Kline. 2013. "Workplace Heterogeneity and the Rise of West German Wage Inequality." *Quarterly Journal of Economics* 128(3): 967–1015.

Card, David, Ana Rute Cardoso, and Patrick M. Kline. 2016. "Bargaining, Sorting, and the Gender Wage Gap: Quantifying the Impact of Firms on the Relative Pay of Women." *Quarterly Journal of Economics* 131(2): 633–86.

Card, David, Ana Rute Cardoso, Joerg Heining, and Patrick Kline. 2018. "Firms and Labor Market Inequality: Evidence and Some Theory." *Journal of Labor Economics* 36(S1): S13–S70.

Carrington, William, J., and Bruce C. Fallick. 2017. "Why Do Earnings Fall with Job Displacement?" *Industrial Relations* 56(4): 688–722.

Couch, Kenneth A., and Dana W. Placzek (CP). 2010. "Earnings Losses of Displaced Workers Revisited." *American Economic Review* 100(1): 572–589.

Davidson, Carl, and Stephen A. Woodbury. 2000. "Wage-Rate Subsidies for Dislocated Workers." *Research in Employment Policy* (2000): 141–184. (Also W.E. Upjohn Institute Staff Working Paper 95-31.)

Davis, Steven J. and Till von Wachter. 2012. "Recessions and the Costs of Job Loss." *Brookings Papers on Economic Activity* 43(2): 1–72.

Dickens, William T., and Lawrence F. Katz. 1986. "Interindustry Wage Differences and Industry Characteristics." In *Unemployment and the Structure of Labor Markets*, Kevin Lang and Jonathan Leonard, eds. Oxford: Basil-Blackwell.

Dickens, Wiliam T., and Lawrence F. Katz. 1987. "Inter-Industry Wage Differences and Theories of Wage Determination." NBER Working Paper 2271.

Fackler, Daniel, Steffen Mueller, and Jens Stegmaier. 2016. "Plant-level Employment Development before Collective Displacement: Comparing Mass Layoffs, Plant Closures, and Bankruptcies." Halle Institute for Economic Research (IWH) Discussion Paper No. 27/2016.

Fackler, Daniel, Steffen Mueller, and Jens Stegmaier. 2017. "Wage Losses after Job Displacement: Productivity Depreciations or Lost Firm Rents?" Unpublished manuscript.

Fallick, Bruce C. 1996. "A Review of the Recent Empirical Literature on Displaced Workers." *Industrial and Labor Relations Review* 50(1): 5–16.

Farber, Henry S. 1993. "The Incidence and Costs of Job Loss, 1982–91." *Brookings Papers on Economic Activity: Microeconomics* 1993(1): 73–132.

Farber, Henry S. 1997. "The Changing Face of Job Loss in the United States, 1981–1995." *Brookings Papers on Economic Activity: Microeconomics* (1997): 55–128.

Farber, Henry S. 2015. "Job Loss in the Great Recession and its Aftermath: U.S. Evidence from the Displaced Workers Survey." NBER Working Paper No. 21216.

Farber, Henry S. 2017. "Employment, Hours, and Earnings Consequences of Job Loss: U.S. Evidence from the Displaced Workers Survey." *Journal of Labor Economics* 35(S1): S235–S272.

Flaaen, Aaron B., Matthew D. Shapiro, and Isaac Sorkin. 2017. "Reconsidering the Consequences of Worker Displacements: Firm versus Worker Perspective," NBER Working Paper No. 24077.

Gibbons, Robert, and Lawrence Katz. 1992. "Does Unmeasured Ability Explain Inter-Industry Wage Differentials." *Review of Economic Studies* 59(3): 515-535.

Goldschmidt, Deborah, and Johannes, F. Schmieder. 2017. "The Rise of Domestic Outsourcing and the Evolution of the German Wage Structure." *Quarterly Journal of Economics*, Forthcoming.

Groshen, Erika, L. 1991. "Sources of Intra-Industry Wage Dispersion: How Much Do Employers Matter?" *Quarterly Journal of Economics* 106(3): 869-884.

Hamermesh, Daniel S. 1996. "What Do We Know About Worker Displacement in the U.S.?" *Industrial Relations* 28(1): 51–59.

Jacobson, Louis S., Robert J. LaLonde, and Daniel G. Sullivan (JLS). 1993a. "Earnings Losses of Displaced Workers." *American Economic Review* 83(4): 685–709.

Jacobson, Louis S., Robert J. LaLonde, and Daniel G. Sullivan (JLS). 1993b. *The Costs of Worker Dislocation* Kalamazoo, MI: W.E. Upjohn Institute for Employment Research.

Jovanovic, Boyan. 1979. "Job Matching and the Theory of Turnover." *Journal of Political Economy* 87(5): 972–990.

Jung, Philip, and Moritz Kuhn. forthcoming. "Earnings Losses and Labor Mobility over the Lifecycle." *Journal of the European Economic Association*.

Kalil, Ariel, and Thomas DeLeire. 2013. "Involuntary Job Transitions and Subjective Well-Being." In *Lifecycle Events and Their Consequences: Job Loss, Family Change, and Declines in Health*, Kenneth A. Couch, Mary C. Daly, and Julie M. Zissimopoulos, eds. Stanford, CA: Stanford University Press.

Katz, Lawrence, and Lawrence Summers. 1989. "Industry Rents: Evidence and Implications." *Brookings Papers on Economic Activity: Microeconomics*: 209–75.

Kletzer, Lori G. 1989. "Returns to Seniority after Permanent Job Loss." *American Economic Review* 79(3):536–43.

Kletzer, Lori G. 1998. "Job Displacement." Journal of Economic Perspectives 12(1): 115-136.

Krolikowski, Pawel. 2018. "Choosing a Control Group for Displaced Workers." *Industrial Labor and Relations Review*, forthcoming.

Krueger, Alan, and Lawrence Summers. 1986. "Reflections on the Inter-Industry Wage Structure." In *Unemployment and the Structure of Labor Markets*, Kevin Lang and Jonathan Leonard, eds. Oxford: Basil-Blackwell.

Krueger, Alan B., and Lawrence H. Summers. 1988. "Efficiency Wages and the Inter-Industry Wage Structure." *Econometrica* 56(2): 259–293.

Lachowska, Marta, Merve Meral, and Stephen A. Woodbury. 2016. "Effects of the Unemployment Insurance Work Test on Long-Term Employment Outcomes." *Labour Economics* 41: 246–265.

Lachowska, Marta, Alexandre Mas, and Stephen A. Woodbury. 2017. "Errors in Self-Reported Work Hours." Manuscript, August.

Murphy, Kevin, and Robert Topel. 1986. "Unemployment, Risk, and Earnings." In *Unemployment and the Structure of Labor Markets*, Kevin Lang and Jonathan Leonard, eds. Oxford: Basil-Blackwell.

Neal, Derek. 1995. "Industry-Specific Human Capital: Evidence from Displaced Workers." *Journal of Labor Economics* 13(4): 653–677.

Oreopoulos, Philip, Marianne Page, and Ann Huff Stevens. 2008. "The Intergenerational Effects of Worker Displacement." *Journal of Labor Economics* 26(3): 455-483.

Parent, Daniel. 2000. "Industry-Specific Capital and the Wage Profile: Evidence from the National Longitudinal Survey of Youth and the Panel Study of Income Dynamics." *Journal of Labor Economics* 18(2):306–23.

Schaller, Jessamyn, and Ann Huff Stevens. 2015. "Short-Run Effect of Job Loss on Health Conditions, Health Insurance, and Health Care Utilization." *Journal of Health Economics* 43: 190–203.

Schoeni, Robert F., and Michael Dardia. 2003. "Estimates of Earnings Losses of Displaced Workers Using California Administrative Data." Research Report No. 03-543, Population Studies Center, University of Michigan.

Song Jae, David Price, Fatih Guvenen, Nick Bloom, and Till von Wachter. 2015. "Firming Up Inequality." NBER Working Paper No. 21199.

Song, Jae, and Till von Wachter. 2014. "Long-Term Nonemployment and Job Displacement." In Federal Reserve Bank of Kansas City, Symposium on Re-Evaluating Labor Market Dynamics < https://www.kansascityfed.org/publications/research/escp>.

Sorkin, Isaac. 2016. "Ranking Firms Using Revealed Preference," Unpublished manuscript.

Stephens, Melvin Jr. 2001. "The Long-Run Consumption Effects of Earnings Shocks." *Review of Economics and Statistics* 83(1): 28–36.

Stevens, Ann Huff. 1997. "Persistent Effects of Job Displacement: The Importance of Multiple Job Losses." *Journal of Labor Economics* 15(1): 165–188.

Stevens, Ann Huff, and Jessamyn Schaller. 2011. "Short-Run Effects of Parental Job Loss on Children's Academic Achievement." *Economics of Education Review* 30(2): 289–299.

Sullivan, Daniel G., and Till von Wachter. 2009. "Job Displacement and Mortality: An Analysis using Administrative Data." *Quarterly Journal of Economics* 124(3): 1265–1306.

Topel, Robert. 1990. "Specific Capital and Unemployment: Measuring the Costs and Consequences of Job Loss." *Carnegie-Rochester Conference Series on Public Policy* 33: 181–214.

Von Wachter, Till. 2010. "Summary of the Literature on Job Displacement in the US and EU: What We Know and What We Would Like to Know." In *Wage Structures, Employment Adjustments and Globalization*, David Marsden and François Rycx, eds. Palgrave Macmillan. Pp. 64–121.

Von Wachter, Til, Elizabeth Weber Handwerker, and Andrew K.G. Hildreth. 2009. "Estimating the 'True' Cost of Job Loss: Evidence Using Matched Data from California, 1991-2000." CES 09-14, Center for Economic Studies, U.S. Census Bureau.

Von Wachter, Till, Jae Song, and Joyce Manchester. 2009. "Long-Term Earnings Losses due to Mass-Layoffs During the 1982 Recession: An Analysis Using Longitudinal Administrative Data from 1974 to 2004." Unpublished manuscript.

Wooldridge, Jeffrey, M. *Econometric Analysis of Cross Section and Panel Data*. Cambridge, MA: MIT Press, 2002.

# Table 1Sample descriptive statistics

	(1)	(2)	(3)	(4)	(5)	(6)
			Workers outside NAICS		Workers employed by top-	
	All	workers	indust	ries 51–56	quintile	employers
-	Displaced	Non-displaced	Displaced	Non-displaced	Displaced	Non-displaced
- Ouarterly average earnings, hours, and wage rates	. 2002–2005					
Earnings (\$)	13,739	12,482	12,417	12,135	15,759	16,243
	(6,668)	(5,996)	(6,215)	(5,490)	(6,457)	(6,042)
Paid work hours	518	500	508	500	537	514
	(80)	(95)	(85)	(97)	(64)	(84)
Hourly wage rate (\$/hours)	64.41	51.22	52.68	49.49	75.15	65.23
	(44.45)	(38.96)	(38.66)	(37.69)	(44.94)	(39.54)
Worker characteristics. 2007:1V						
Female (proportion)	0.300	0.359	0.274	0.353	0.240	0.230
Race (proportions)						
White, not Hispanic	0.774	0.677	0.736	0.671	0.813	0.739
Black, not Hispanic	0.031	0.033	0.030	0.032	0.022	0.034
Hispanic	0.074	0.133	0.105	0.139	0.043	0.058
Asian/Pacific Islander	0.071	0.100	0.078	0.100	0.068	0.109
American Indian or Alaskan Native	0.014	0.014	0.015	0.015	0.012	0.011
Missing, unknown, or not available	0.037	0.044	0.035	0.044	0.042	0.050
Schooling (proportions)						
less than high school	0.079	0.129	0.111	0.135	0.053	0.062
GED	0.031	0.032	0.038	0.033	0.022	0.035
high school graduate	0.446	0.462	0.469	0.472	0.461	0.496
some college	0.153	0.161	0.151	0.158	0.134	0.183
associate's degree	0.125	0.101	0.121	0.102	0.125	0.108
bachelors degree	0.135	0.093	0.095	0.081	0.161	0.093
master's/PhD	0.030	0.023	0.016	0.019	0.043	0.022
Age (years)	39.68	41.47	39.01	41.53	40.53	41.87
	(6.39)	(6.44)	(6.53)	(6.44)	(5.90)	(6.20)
Employer characteristics in 2007:IV						
Employer size (number of workers)	2,097	8,478	824	8,400	2,881	16,622
	(2,562)	(20,065)	(1,550)	(20,317)	(2,694)	(28,378)
NAICS Industry (proportions)						
11 agriculture, forestry, fishing	0.014	0.045	0.024	0.048	0.001	0.002
21–23 mining, utilities, construction	0.078	0.082	0.132	0.087	0.097	0.153
31–33 manufacturing	0.268	0.460	0.455	0.491	0.274	0.631
42–49 trade, transportation	0.152	0.145	0.258	0.155	0.035	0.058
51–56 information, finance, prof. services	0.411	0.063	n/a	n/a	0.589	0.103
61–62 educational and health care services	0.013	0.093	0.021	0.099	0.003	0.024
/1-/2 arts, recreation, hospitality services	0.051	0.032	0.087	0.034	n/a	0.000
81 other services	0.007	0.006	0.011	0.006	0.001	0.002
92-99 public auministration and unclassified	0.007	0.075	0.012	0.080	0.001	0.028
Number of employers (pre- and post-displacement)	3,621	1,570	513	1,383	195	447
Number of workers	3,032	13,290	1,786	12,447	1,802	5,621

Notes : Earnings are expressed in constant 2010 dollars. Standard deviations are in parentheses.

Source: Authors' tabulations of Washington administrative wage and claims records. See section 2.1 for details of the sample construction.

#### Table 2

Summary of estimated losses due to displacement

	Earnings		Но	Hours		wage rate
	Q1	Q17-Q20	Q1	Q17-Q20	Q1	Q17-Q20
All workers						
A. Full losses						
levels	-\$5,964	-\$1,941	-200	-29.3	-1.89	-1.46
	(-47.8%)	(-15.6%)	(-40.0%)	(-5.9%)	(-3.7%)	(-2.9%)
logs	-0.488	-0.152	-0.384	-0.070	-0.109	-0.080
B. Losses attributable to foregone employer fixed effects						
logs	-0.032	-0.017	-0.015	0.003	-0.017	-0.020
Workers by in	dustry group					
C. Full losses,	all workers ex	cept NAICS 51	1–56			
logs	-0.595	-0.206	-0.425	-0.098	-0.172	-0.110
D. Full losses,	NAICS 51-56	workers only				
logs	-0.363	-0.172	-0.384	-0.065	-0.072†	-0.112
Workers displa	aced from emp	oloyers paying	top-quintile ea	rnings premiun	ns	
E. Full losses						
logs	-0.368	-0.125	-0.303	-0.056	-0.074	-0.063
F. Losses attrib	outable to fore	gone employer	fixed effects			
logs	-0.074	-0.067	-0.038	-0.021	-0.036	-0.046

*Note*: Each entry gives the estimated displacement effect on the indicated outcome in either quarter 1 (Q1) or the average of quarters 17, 18, 19, and 20 (Q17-Q20) following displacement. For levels, implied percentage changes relative to the non-displaced comparison group are shown in parentheses. (For example, the estimated effect of displacement on average earnings in quarters 17–20 after displacement is –\$1,941, which is 15.6% less than the earnings of the non-displaced comparison group.)

†Average of Q1 and Q2.

Study	State and time period	Sample	Earnings losses in the first year	Earnings losses after five years
Jacobson, Lalonde, and Sullivan (1993b)	Pennsylvania, 1974–1986	All UI claimants	\$7,800 (66%*)	\$2,900 or (24%*)
Couch and Placzek (2010)	Connecticut, 1993–2004	All UI claimants	\$7,700 (49%)	\$5,100 (32%)
Lachowska, Mas, and Woodbury (2017)	Washington, 2002–2014	All UI claimants	\$5,960 or 48 percent (=\$5,960/\$12,482)	\$1,940 (16%)

Table 3: Estimated earnings losses due to displacement, selected studies using UI administrative records

Sources: Estimates from the papers cited.

\*Percentage estimates for Jacobson, LaLonde, and Sullivan (1993b) are reported in Couch and Placzek (2010), Table 1.

Table 4
Estimated displacement effects on the distribution on average weekly work hours

Quarter relative to					Outcome				
displacement	Hours > 0	Hours $> 10$	Hours > 20	Hours > 30	Hours > 35	Hours > 40	Hours > 45	Hours > 50	Hours > 60
0	-0.065	-0.154	-0.253	-0.360	-0.372	-0.276	-0.057	-0.001	0.005
	(0.005)	(0.007)	(0.009)	(0.010)	(0.010)	(0.009)	(0.007)	(0.005)	(0.002)
1	-0.228	-0.287	-0.346	-0.442	-0.513	-0.291	-0.086	-0.017	0.004
	(0.008)	(0.009)	(0.009)	(0.010)	(0.010)	(0.009)	(0.006)	(0.004)	(0.002)
2	-0.139	-0.195	-0.248	-0.277	-0.240	-0.210	-0.057	-0.005	0.007
	(0.007)	(0.008)	(0.009)	(0.010)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
3	-0.108	-0.158	-0.198	-0.219	-0.166	-0.086	-0.061	-0.011	0.006
	(0.006)	(0.008)	(0.009)	(0.010)	(0.010)	(0.010)	(0.007)	(0.004)	(0.002)
4	-0.095	-0.148	-0.192	-0.196	-0.207	-0.161	-0.040	-0.001	0.009
	(0.006)	(0.007)	(0.008)	(0.009)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
5	-0.081	-0.118	-0.155	-0.145	-0.110	-0.089	-0.014	0.013	0.011
	(0.006)	(0.007)	(0.008)	(0.009)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
6	-0.052	-0.077	-0.111	-0.123	-0.107	-0.112	-0.031	0.002	0.010
	(0.005)	(0.006)	(0.007)	(0.009)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
7	-0.037	-0.068	-0.100	-0.105	-0.101	-0.127	-0.042	0.001	0.009
	(0.005)	(0.006)	(0.007)	(0.009)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
8	-0.038	-0.064	-0.098	-0.114	-0.094	-0.107	-0.034	0.002	0.010
	(0.005)	(0.006)	(0.007)	(0.009)	(0.010)	(0.010)	(0.007)	(0.005)	(0.002)
9	-0.032	-0.058	-0.085	-0.105	-0.111	-0.063	-0.007	0.013	0.012
	(0.004)	(0.006)	(0.007)	(0.008)	(0.010)	(0.010)	(0.008)	(0.006)	(0.003)
10	-0.033	-0.052	-0.076	-0.092	-0.102	-0.065	-0.026	-0.000	0.008
	(0.004)	(0.006)	(0.007)	(0.008)	(0.010)	(0.011)	(0.008)	(0.005)	(0.003)
11	-0.032	-0.044	-0.077	-0.112	-0.101	-0.072	-0.028	0.008	0.007
	(0.004)	(0.005)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.002)
12	-0.033	-0.061	-0.096	-0.110	-0.091	-0.019	0.024	0.029	0.012
	(0.004)	(0.006)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
13	-0.028	-0.051	-0.080	-0.111	-0.121	-0.084	-0.015	0.020	0.018
	(0.004)	(0.005)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
14	-0.034	-0.061	-0.086	-0.095	-0.072	-0.015	0.015	0.026	0.008
	(0.004)	(0.006)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
15	-0.026	-0.047	-0.074	-0.096	-0.096	-0.084	-0.013	0.012	0.011
	(0.004)	(0.005)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
16	-0.027	-0.051	-0.086	-0.093	-0.077	-0.020	0.010	0.017	0.006
	(0.004)	(0.005)	(0.007)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
17	-0.026	-0.047	-0.074	-0.080	-0.087	-0.057	0.002	0.022	0.020
	(0.004)	(0.005)	(0.006)	(0.008)	(0.009)	(0.011)	(0.008)	(0.006)	(0.003)
18	-0.026	-0.048	-0.062	-0.066	-0.065	-0.028	0.023	0.028	0.012
	(0.004)	(0.005)	(0.006)	(0.008)	(0.009)	(0.011)	(0.009)	(0.006)	(0.003)
19	-0.022	-0.037	-0.056	-0.073	-0.089	-0.085	-0.028	0.008	0.018
	(0.004)	(0.005)	(0.006)	(0.008)	(0, 010)	(0.011)	(0, 009)	(0.006)	(0.003)
20	-0.017	-0.040	-0.075	-0.085	-0.084	-0.054	-0.008	0.015	0.008
20	(0.004)	(0.005)	(0.007)	(0.008)	(0.010)	(0.012)	(0,009)	(0.006)	(0.003)
	(0.001)	(0.000)	(0.007)	(0.000)	(0.010)	(0.012)	(0.00))	(0.000)	(0.005)
Comparison group mean	0.996	0.980	0.952	0.867	0.758	0.457	0.207	0.095	0.022
Worker_quarter									
observations	826 219	826 219	826 219	826 219	826 219	826 219	826 219	826 219	826 219
Number of	020,217	020,217	020,217	020,217	020,217	020,217	020,217	020,217	020,217
workers	16 222	16 222	16 222	16 222	16 222	16 222	16 222	16 222	16 222
WOIK015	10,322	10,322	10,322	10,322	10,322	10,322	10,322	10,322	10,322

*Notes*: Each column shows estimated changes in the probability of working at least a given number of weekly hours (on average over a quarter), based on estimates of equation (1). For example, the column headed "Hours > 35" is based on equation (1) with an indicator of whether the worker's weekly hours exceeded 35 as the dependent variable. Standard errors clustered by worker are shown in parentheses.

Table 5 Estimated displacement effects on log earnings, log hours, log wages, and AKM employer fixed effects ( $\psi$ )

	Log earnings	ψ log earnings	Log hours	ψ log hours	Log wage	ψ log wage
Quarter relative to displacement	(1)	(2)	(3)	(4)	(5)	(6)
0	-0.36***	0.00*	-0.40***	0.00***	0.06***	-0.00
	(0.02)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
1	-0.49***	-0.03***	-0.38***	-0.01***	-0.11***	-0.02***
	(0.02)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
2	-0.39***	-0.03***	-0.25***	-0.01***	-0.15***	-0.02***
	(0.02)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
3	-0.31***	-0.03***	-0.20***	-0.01***	-0.12***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
4	-0.32***	-0.03***	-0.22***	-0.01***	-0.11***	-0.02***
	(0.02)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
5	-0.26***	-0.03***	-0.13***	-0.01***	-0.13***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
6	-0.23***	-0.02***	-0.11***	-0.01**	-0.11***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
7	-0.24***	-0.02***	-0.13***	-0.01***	-0.11***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
8	-0.23***	-0.02***	-0.11***	-0.01**	-0.12***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
9	-0.22***	-0.02***	-0.11***	-0.01**	-0.11***	-0.01***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
10	-0.19***	-0.02***	-0.09***	-0.00	-0.09***	-0.01***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
11	-0.22***	-0.01***	-0.09***	0.00	-0.12***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
12	-0.18***	-0.01***	-0.09***	0.01**	-0.09***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
13	-0.19***	-0.01***	-0.09***	0.01*	-0.10***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
14	-0.15***	-0.01***	-0.08***	0.01**	-0.0/***	-0.02***
1.5	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
15	-0.18***	-0.01**	-0.08***	0.01**	-0.09***	-0.02***
16	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
16	-0.1/***	-0.01***	-0.08***	0.01**	-0.09***	-0.02***
17	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
1 /	-0.1/***	-0.01***	-0.08***	0.01*	-0.09***	-0.02***
19	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
18	-0.13	-0.02***	-0.03	(0.00)	-0.07***	-0.02***
10	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
17	-0.10	-0.02	-0.00	(0.00)	-0.09	-0.02
20	-0.15***	-0.02***	-0.02***	0.00)	-0.07***	-0.02***
20	-0.13	-0.02	-0.08	(0.00)	-0.07***	-0.02***
	(0.01)	(0.00)	(0.01)	(0.00)	(0.01)	(0.00)
Number of worker-quarters	822,933	822,279	819,241	822,279	819,233	822,279
Number of workers	16,322	16,322	16,322	16,322	16,322	16,322
$R^2$	0.083	0.030	0.054	0.019	0.066	0.029

*Notes*: Estimates in columns (1), (3), and (5) are based on equation (1) and are graphed in Figures 2, 3, and 4. Estimates in columns (2), (4), and (6) are based on equation (4) and are graphed in Figure 10. For brevity, estimated coefficients for quarters –20 to –1 relative to displacement are not displayed. Each regression also controls for a worker-specific fixed effect; a vector of quarterly dummies; worker's age and age squared; a vector of gender, race, and education dummies interacted with the worker's age; logarithm of pre-displacement employer size and 1-digit NAICS code in 2007:IV interacted with a vector of yearly dummies; a simple average of pre-displacement earnings with the primary employer and an average of pre-displacement hours with the primary employer, each interacted with a vector of yearly dummies. Earnings are expressed in constant 2010 dollars.

Standard errors clustered by worker in parentheses (\*\*\* p < 0.01; \*\* p < 0.05; \*p < 0.1).



Figure 1 Earnings (top) and hours (bottom) profiles of displaced and non-displaced workers

*Notes*: The top figure shows the quarterly earnings profiles (constant 2010 dollars) of workers displaced in Washington during the first quarter of 2009 (solid blue line) and the non-displaced comparison group (dashed red line). The bottom figure shows the work hour profiles of the same two groups. Both earnings and hours are unconditional (that is, include observations of zero earnings and hours). The vertical lines denote the quarter of separation.





*Notes*: The top figure shows estimated  $\delta_k$ s—quarterly unconditional earnings lost due to displacement (in constant 2010 \$1,000s)—based on equation (1) with unconditional earnings from the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly earnings lost due to displacement—based on equation (1) with the log of earnings from the primary employer as the dependent variable. Whiskers (which are very small) denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.



Figure 3 Estimated work hour losses due to displacement, Washington, 2008–2010

*Notes*: The top figure shows estimated  $\delta_k$ s—quarterly unconditional hours lost due to displacement—based on equation (1) with unconditional hours at the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly hours lost due to displacement—based on equation (1) with the log of hours at the primary employer as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.





*Notes*: The figure shows estimated  $\delta_k$ s—the reduction in the log hourly wage rate due to displacement—based on equation (1) with the log of hourly wage rate at the primary employer (constant 2010 dollars per hour) as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

## Figure 5 Estimated displacement effects for workers displaced from industries other than NAICS industries 51–56



*Notes*: The figures show estimated displacement effects for workers displaced from any industry *except* NAICS industries 51–56 (information, finance and insurance, real estate, professional, scientific, and technical services, management of companies; administrative, support, and waste management and remediation services). Each figure shows the profile of displacement effects for an outcome—log quarterly earnings, log quarterly hours, and log wage rate (all from the primary employer), or the probability of employment (positive earnings or hours)—based on estimates of  $\delta_k$  in equation (1). Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

Figure 6 Estimated displacement effects by job tenure at time of displacement



*Notes*: The figures show estimated displacement effects for workers with 3–4 years of job tenure at the time of displacement (red circles), and 6 or more years of job tenure at the time of displacement (blue dots, repeated from Figures 2, 3, 4 for the first three panels). The omitted (reference) time period for workers displaced with 3–4 years of tenure (and their comparison group) is 3 years before displacement. Each figure shows the profile of displacement effects for an outcome—quarterly log earnings, log hours, log wage rate (all from the primary employer), or the probability of employment (positive earnings or hours)—based on estimates of  $\delta_k$  in equation (1). Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.





*Notes*: The figures show estimated displacement effects for a sample of displaced workers not required to be observed with positive earnings or hours after being displaced (blue dots), and for the sample used in Figures 2, 3, and 4 (observed with positive earnings or hours in at least one quarter of each year following displacement—red circles, repeated from Figures 2, 3, and 4). The whiskers denote 95–percent confidence intervals clustered by worker. The vertical lines denote the quarter of displacement.





*Notes*: The figures show estimated displacement losses attributable to foregone employer fixed effects (blue dots) compared with the full losses due to displacement (red circles, repeated from Figures 2, 3, and 4). Losses attributable to foregone employer fixed effects are estimates of  $\delta_k$  from equation (4). For example, to obtain the estimates of earnings lost due to foregone employer premiums, equation (4) was estimated with the AKM employer fixed effect ( $\hat{\psi}$ ) for log earnings as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement. *Source*: Authors' calculations using Washington administrative wage and claims records. See sections 3.2 and 5.1 of the text for details.

#### Figure 9

Estimated displacement losses due to foregone employer fixed effects for workers displaced from employers paying top-quintile earnings premiums



*Notes*: The figures show estimated displacement losses attributable to foregone employer fixed effects (blue dots, estimated from equation (4)), and full losses due to displacement (red circles, estimated from equation (1)), for workers displaced from employers paying top-quintile earnings premiums (60 percent of displaced workers in the sample—see columns 5 and 6 of Table 1). Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

#### Figure 10

Estimated displacement losses due to foregone employer fixed effects for workers displaced from employers paying earnings premiums below the top quintile



*Notes*: The figures show estimated displacement losses attributable to foregone employer premiums (solid blue dots, estimated from equation (4)), and full losses due to displacement (red circles, estimated from equation (1)) and for workers displaced from employers paying an earnings premium below the top quintile (40 percent of displaced workers in the sample). Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

#### Figure 11

Losses of displaced workers who moved to employers paying the same, lower, and higher earnings premiums compared with the separating employer



*Notes*: The figures show the losses of displaced workers who, eight quarters after separation, were reemployed by an employer paying an earnings premium in the same quintile as the separating employer (blue dots, 65.1 percent of the sample), by an employer paying a lower-quintile earnings premium than the separating employer (red circles, 18.4 percent of the sample), and by an employer paying a higher-quintile earnings premium than the separating employer (green squares, 16.5 percent of the sample). Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

Figure 12 Estimated displacement losses due to foregone industry fixed effects



*Notes*: The figures show estimated displacement losses attributable to foregone industry fixed effects (blue dots) compared with the full losses due to displacement (red circles). For example, to obtain the estimates of lost earnings due to foregone industry fixed effects, equation (4) was estimated with the AKM industry fixed effect ( $\hat{\lambda}_k$ ) for log earnings as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

### Sources of Displaced Workers' Long-Term Earnings Losses: Appendices

Marta Lachowska Alexandre Mas Stephen A. Woodbury

December 31, 2017

Appendix A: Additional tables and figures

Appendix B: Estimation of employer fixed effects for earnings, hours, and wage rates

Appendix C: Estimates using an alternative comparison group

#### Appendix A: Additional tables and figures

Appendix Table A.1 Estimated effects of displacement on unconditional earnings and log earnings (from primary employer and all employers)

		All earnings (in		
Outcome variable	Earnings (in \$1,000s)	\$1,000s)	Log earnings	earnings
	(1)	(2)	(3)	(4)
Covariates			~ /	
Ouarter since displacement				
-20	0 21***	0 21***	0 01***	0 01***
_ •	(0, 06)	(0.06)	(0, 00)	(0,00)
_19	0 14**	0 14**	0.00	0.00
	(0, 06)	(0.06)	(0,00)	(0,00)
-18	0.72***	0.73***	0.05***	0.05***
-	(0.07)	(0.07)	(0.00)	(0.00)
-17	0.25***	0.25***	0.01**	0.01**
	(0.07)	(0.07)	(0.00)	(0.00)
-16	0.31***	0.31***	0.02***	0.02***
	(0.07)	(0.07)	(0.00)	(0.00)
-15	0.52***	0.53***	0.02***	0.02***
10	(0, 08)	(0.09)	(0.01)	(0.01)
-14	0.68***	0.68***	0.04***	0.04***
	(0, 08)	(0.08)	(0.00)	(0, 00)
-13	0.61***	0.61***	0.03***	0.03***
	(0.08)	(0.08)	(0.01)	(0.01)
-12	-0.16**	-0.16**	-0.02***	-0.02***
	(0.08)	(0.08)	(0.01)	(0.01)
-11	0 46***	0 47***	0.02***	0.02***
	(0, 09)	(0, 09)	(0.01)	(0.01)
-10	-0.01	0.01	-0.01	-0.01
	(0.08)	(0.08)	(0.01)	(0.01)
_9	0 48***	0 49***	0.03***	0.03***
	(0.08)	(0.09)	(0.01)	(0.01)
-8	-0.35***	-0.34***	-0.03***	-0.03***
-	(0.08)	(0.08)	(0.01)	(0.01)
-7	0.36***	0.38***	0.01	0.01
	(0.09)	(0.09)	(0.01)	(0.01)
6	-0.13	-0.11	-0.01**	-0.01**
	(0.09)	(0.09)	(0.01)	(0.01)
-5	0.12	0.14	0.01	0.01*
	(0.09)	(0.09)	(0.01)	(0.01)
-4	-0.77***	-0.75***	-0.06***	-0.05***
	(0.09)	(0.09)	(0.01)	(0.01)
-3	-0.17*	-0.14	-0.03***	-0.03***
	(0.10)	(0.10)	(0.01)	(0.01)
-2	-0.51***	-0.49***	-0.06***	-0.05***
	(0.10)	(0.10)	(0.01)	(0.01)
-1	-0.01	0.03	-0.02***	-0.01*
	(0.11)	(0.11)	(0.01)	(0.01)
0	-1.91***	-0.94***	-0.36***	-0.27***
-	(0.17)	(0.18)	(0.02)	(0.02)
1	-5.96***	-5.16***	-0.49***	-0.39***
	(0.16)	(0.17)	(0.02)	(0.02)
	()	()	(=)	(

2	-4.44***	-4.22***	-0.39***	-0.36***
	(0.15)	(0.15)	(0.02)	(0.02)
3	-3.42***	-3.23***	-0.31***	-0.28***
	(0.14)	(0.14)	(0.01)	(0.01)
4	-3.22***	-3.04***	-0.32***	-0.30***
	(0.15)	(0.15)	(0.02)	(0.01)
5	-2.82***	-2.65***	-0.26***	-0.23***
	(0.14)	(0.14)	(0.01)	(0.01)
6	-2.48***	-2.34***	-0.23***	-0.21***
	(0.14)	(0.14)	(0.01)	(0.01)
7	-2.49***	-2.37***	-0.24***	-0.22***
	(0.13)	(0.13)	(0.01)	(0.01)
8	-2.53***	-2.40***	-0.23***	-0.21***
	(0.13)	(0.13)	(0.01)	(0.01)
9	-2.06***	-1.90***	-0.22***	-0.20***
	(0.14)	(0.14)	(0.01)	(0.01)
10	-2.08***	-1.94***	-0.19***	-0.16***
	(0.14)	(0.14)	(0.01)	(0.01)
11	-2.46***	-2.07***	-0.22***	-0.18***
	(0.13)	(0.13)	(0.01)	(0.01)
12	-2.03***	-1.92***	-0.18***	-0.17***
	(0.13)	(0.13)	(0.01)	(0.01)
13	-2.09***	-1.89***	-0.19***	-0.17***
	(0.14)	(0.14)	(0.01)	(0.01)
14	-1.73***	-1.54***	-0.15***	-0.13***
	(0.13)	(0.14)	(0.01)	(0.01)
15	-2.12***	-1.93***	-0.18***	-0.16***
	(0.13)	(0.14)	(0.01)	(0.01)
16	-2.00***	-1.86***	-0.17***	-0.15***
17	(0.14)	(0.14)	(0.01)	(0.01)
	-2.00***	-1.81***	-0.17***	-0.15***
10	(0.14)	(0.14)	(0.01)	(0.01)
18	-1./1***	-1.36***	-0.13***	-0.11***
10	(0.14)	(0.14)	(0.01)	(0.01)
19	-2.12***	-1.99***	-0.16***	-0.14***
20	(0.14)	(0.14)	(0.01)	(0.01)
20	$-1.94^{+++}$	$-1.81^{+++}$	$-0.15^{***}$	-0.14***
N	(0.14)	(0.14)	(0.01)	(0.01)
Number of quarter-year	926 210	006 010	800 000	000 000
workers	826,219	826,219	822,933	822,933
Number of workers	16,322	16,322	16,322	16,322
K-squarea	0.091	0.087	0.083	0.078

*Notes*: Estimates in columns 1 and 3 are based on equation (1) and are graphed in Figure 2. The sample consists of displaced workers and the non-displaced comparison group who were UI claimants at some time during 2002–2014 (see section 2.1 of the text for details). Each regression also controls for a worker-specific fixed effect; a vector of quarterly dummies; worker's age and age squared; a vector of gender, race, and education dummies interacted with the worker's age; logarithm of predisplacement employer size and 1-digit NAICS code in 2007:IV interacted with a vector of yearly dummies; a simple average of pre-displacement earnings with the primary employer and an average of pre-displacement hours with the primary employer, each interacted with a vector of yearly dummies. Earnings are expressed in constant 2010 dollars. Standard errors clustered by worker in parentheses (\*\*\* p < 0.01; \*\* p < 0.05; \* p < 0.1).

Appendix Table A.2 Estimated effects of displa and all employers)	cement on unconditiona	al hours and	log hours (from prin	nary employer
Outcome variable	Primary hours	All hours	Log primary hours	Log all hours
	(1)	(2)	(3)	(4)

Outcome variable	Primary hours	All hours	Log primary hours	Log all hours
	(1)	(2)	(3)	(4)
Covariates				
Quarter since displacement				
-20	5.77***	5.55***	0.01	0.01
	(1.59)	(1.68)	(0.00)	(0.00)
-19	-3.81**	-3.75**	-0.01	-0.01
	(1.60)	(1.67)	(0.00)	(0.00)
-18	15.86***	16.63***	0.03***	0.03***
	(1.83)	(1.91)	(0.01)	(0.01)
-17	-0.06	0.47	0.00	0.00
	(1.89)	(1.97)	(0.01)	(0.01)
-16	8.09***	8.65***	0.02***	0.02***
	(1.97)	(2.07)	(0.01)	(0.01)
-15	9.02***	10.13***	0.02***	0.02***
	(1.87)	(1.96)	(0.01)	(0.01)
-14	21.55***	23.08***	0.04***	0.05***
	(1.99)	(2.10)	(0.01)	(0.01)
-13	21.62***	22.39***	0.04***	0.04***
	(2.09)	(2.18)	(0.01)	(0.01)
-12	-1.19	-0.07	-0.01	-0.01
	(2.02)	(2.12)	(0.01)	(0.01)
-11	22.23***	23.74***	0.04***	0.04***
	(2.15)	(2.27)	(0.01)	(0.01)
-10	1.78	2.89	0.00	0.00
	(2.09)	(2.23)	(0.01)	(0.01)
-9	21.68***	23.94***	0.04***	0.04***
	(2.21)	(2.35)	(0.01)	(0.01)
-8	7.81***	10.65***	0.01	0.01**
	(2.19)	(2.35)	(0.01)	(0.01)
-7	6.85***	8.37***	-0.00	-0.00
	(2.28)	(2.41)	(0.01)	(0.01)
-6	-1.14	0.57	-0.01	-0.01
	(2.21)	(2.34)	(0.01)	(0.01)
-5	5.55**	6.39***	0.00	0.00
	(2.30)	(2.42)	(0.01)	(0.01)
-4	-8.60***	-7.18***	-0.03***	-0.03***
	(2.32)	(2.47)	(0.01)	(0.01)
-3	-8.83***	-6.65**	-0.04***	-0.04***
	(2.47)	(2.62)	(0.01)	(0.01)
-2	-15.34***	-13.96***	-0.04***	-0.03***
	(2.28)	(2.41)	(0.01)	(0.01)
-1	-5.09*	-4.11	-0.01	-0.01*
	(2.65)	(2.77)	(0.01)	(0.01)
0	-136.26***	-98.41***	-0.40***	-0.30***
	(3.66)	(3.80)	(0.01)	(0.01)
1	-199.82***	-176.32***	-0.38***	-0.31***
	(4.14)	(4.48)	(0.01)	(0.01)
2	-123.86***	-115.82***	-0.25***	-0.22***
	(4.20)	(4.31)	(0.01)	(0.01)
3	-94.46***	-86.91***	-0.20***	-0.18***
	(4.04)	(4.16)	(0.01)	(0.01)

4	-95.53***	-89.43***	-0.22***	-0.20***
	(3.93)	(4.04)	(0.01)	(0.01)
5	-63.99***	-57.80***	-0.13***	-0.11***
	(3.82)	(3.91)	(0.01)	(0.01)
6	-53.05***	-48.96***	-0.11***	-0.10***
	(3.57)	(3.65)	(0.01)	(0.01)
7	-50.01***	-45.27***	-0.13***	-0.11***
	(3.46)	(3.58)	(0.01)	(0.01)
8	-46.73***	-41.56***	-0.11***	-0.10***
	(3.43)	(3.53)	(0.01)	(0.01)
9	-40.38***	-35.17***	-0.11***	-0.09***
	(3.36)	(3.49)	(0.01)	(0.01)
10	-39.64***	-33.85***	-0.09***	-0.07***
	(3.29)	(3.39)	(0.01)	(0.01)
11	-41.38***	-27.34***	-0.09***	-0.06***
	(3.27)	(3.44)	(0.01)	(0.01)
12	-35.30***	-31.40***	-0.09***	-0.08***
	(3.34)	(3.46)	(0.01)	(0.01)
13	-40.42***	-32.50***	-0.09***	-0.07***
	(3.27)	(3.38)	(0.01)	(0.01)
14	-32.43***	-26.83***	-0.08***	-0.07***
	(3.35)	(3.43)	(0.01)	(0.01)
15	-36.85***	-30.65***	-0.08***	-0.07***
	(3.21)	(3.30)	(0.01)	(0.01)
16	-32.47***	-26.97***	-0.08***	-0.07***
	(3.27)	(3.35)	(0.01)	(0.01)
17	-31.41***	-23.75***	-0.08***	-0.06***
	(3.31)	(3.42)	(0.01)	(0.01)
18	-23.48***	-18.87***	-0.05***	-0.05***
	(3.30)	(3.40)	(0.01)	(0.01)
19	-30.53***	-25.04***	-0.06***	-0.05***
	(3.32)	(3.44)	(0.01)	(0.01)
20	-31.87***	-26.35***	-0.08***	-0.07***
	(3.41)	(3.48)	(0.01)	(0.01)
Number of quarter-year workers	826,219	826,219	819,241	819,500
Number of workers	16,322	16,322	16,322	16,322
R-squared	0.114	0.101	0.054	0.049

*Notes:* Estimates in columns 1 and 3 are based on equation (1) and are graphed in Figure 3. The sample consists of displaced workers and the non-displaced comparison group who were UI claimants at some time during 2002–2014 (see section 2.1 of the text for details). Each regression also controls for a worker-specific fixed effect; a vector of quarterly dummies; worker's age and age squared; a vector of gender, race, and education dummies interacted with the worker's age; logarithm of pre-displacement employer size and 1-digit NAICS code in 2007:IV interacted with a vector of yearly dummies; a simple average of pre-displacement hours with the primary employer, each interacted with a vector of yearly dummies.

Standard errors clustered by workers are in parentheses (\*\*\* p < 0.01; \*\* p < 0.05; \* p < 0.1).

Appendix Figure A.1





*Notes*: The top figure shows estimated  $\delta_k$ s—quarterly unconditional earnings lost due to displacement (in constant 2010 \$1,000s)—based on the random trend model (equation (2)) with unconditional earnings from the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly earnings lost due to displacement—based on equation (2) with the log of earnings from the primary employer as the dependent variable. Whiskers (which are very small) denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement. *Source*: Authors' calculations using Washington administrative wage and claims records. See sections 3.1, 4.1, and 4.2 of the text for details.

Appendix Figure A.2

Estimated work hour losses due to displacement, Washington, 2008–2010, based on the random-trend model



*Notes*: The top figure shows estimated  $\delta_k$ s—quarterly unconditional hours lost due to displacement—based on the random trend model (equation (2)) with unconditional hours at the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly hours lost due to displacement—based on equation (2) with the log of hours at the primary employer as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

Appendix Figure A.3

Estimated hourly wage rate losses due to displacement, Washington, 2008–2010, based on the random-trend model



*Notes*: The figure shows estimated  $\delta_k$ s—the reduction in the log hourly wage rate due to displacement—based on the random trend model (equation (2)) with the log of the hourly wage rate at the primary employer (constant 2010 dollars per hour) as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.
Estimated earnings losses due to displacement, based on the broadened sample of displaced and non-displaced workers, Washington, 2008–2010



*Notes*: The figures show earnings losses estimated using the broadened sample of displaced and non-displaced workers—that is, without restricting the sample to workers who claimed UI at some time during 2002–2014—see the discussion in section 2.1. (For estimates using the UI claimant sample, see Figure 2 in the main text.) The top figure shows estimated  $\delta_k$ s—quarterly unconditional earnings lost due to displacement (in constant 2010 \$1,000s)—based on equation (1) with unconditional earnings from the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly earnings lost due to displacement—based on equation (1) with the log of earnings from the primary employer as the dependent variable. Whiskers (which are very small) denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement. *Source*: Authors' calculations using Washington administrative wage and claims records. See section 2.1 of the text for details of the broadened and UI claimant samples.

Estimated work hour losses due to displacement, based on the broadened sample of displaced and non-displaced workers, Washington, 2008–2010



*Notes*: The figures show quarterly work hour losses estimated using the broadened sample of displaced and non-displaced workers—that is, without restricting the sample to workers who claimed UI at some time during 2002–2014—see the discussion in section 2.1. (For estimates using the UI claimant sample, see Figure 3 in the main text.) The top figure shows estimated  $\delta_k$ s—quarterly unconditional hours lost due to displacement—based on equation (1) with unconditional hours at the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly hours lost due to displacement—based on equation (1) with the log of hours at the primary employer as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See section 2.1 of the text for details of the broadened and UI claimant samples.

Estimated hourly wage rate losses due to displacement, based on the broadened sample of displaced and non-displaced workers, Washington, 2008–2010



*Notes*: The figure shows hourly wage rate losses estimated using the broadened sample of displaced and non-displaced workers—that is, without restricting the sample to workers who claimed UI at some time during 2002–2014—see the discussion in section 2.1. (For estimates using the UI claimant sample, see Figure 4 in the main text.) The figure plots estimated  $\delta_k$ s—the reduction in the log hourly wage rate due to displacement—based on equation (1) with the log of the hourly wage rate at the primary employer (constant 2010 dollars per hour) as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See section 2.1 of the text for details of the broadened and UI claimant samples.

# Appendix B: Estimation of employer fixed effects for earnings, hours, and wage rates *B.1 Construction of the analysis sample*

The raw data for the AKM analysis come from quarterly administrative wage records of Washington State for calendar years 2002–2014. These quarterly reports are submitted by all UI– covered employers for the purposes of administering UI benefit claims and payroll taxes. A record appears for each employer–worker–quarter combination, and each record includes a year– quarter indicator; the ID and NAICS industry code of the reporting employer; and the worker ID, earnings, and work hours of the worker with that employer in the specified quarter. The availability of quarterly hours for each employer allows us to include both full–time and part– time workers (and jobs) in the analysis. A worker has as many wage records as he or she has employers in a given quarter.

From these raw data, we construct a linked employer-employee panel for analysis following a procedure developed by Sorkin (2016). We first select each worker's primary employer for each quarter. In most cases, a worker has only one employer during the quarter, but for about 27 percent of the worker–quarter observations, we need to define the worker's primary employer as the employer from whom the worker earned the largest share of his/her earnings in that quarter.

We next define an employment spell as a series of at least five consecutive quarters during which a worker has earnings from the same primary employer. For each of these spells, we drop the first quarter (so as to avoid making inferences about earnings and hours based on a partial quarter of employment) and the last two quarters (to avoid making inferences based on earnings and hours in the quarter before a job loss and the quarter of a job loss).

We then annualize the remaining quarterly data within each calendar year, conditional on the calendar year including at least two consecutive quarters of earnings from the same primary

employer. Earnings are defined as annualized earnings in a given year with the principal employer, and similarly for hours and wage rates. Figure B.1 illustrates the process and gives some examples, described in the figure notes. Ultimately, the unit of observation is the worker–year, with a focus on the primary employer in a year.<sup>1</sup>

Finally, we impose several restrictions on the sample, dropping the following:

- workers with more than 9 employers in a year (this affects 1 percent of the sample)
- workers with annual earnings less than \$2,850 (in 2005 dollars) and workers with calculated hourly wage rates  $\leq$  \$2.00/hour (in 2005 dollars) (Sorkin 2016; Card, Heining, and Kline 2013)
- workers who worked fewer than 400 hours in the year<sup>2</sup>
- workers who worked more than 4,800 hours in the year
- employers with fewer than 5 employees in the year (Song et al. 2015)
- all displaced workers and all non-displaced comparison workers (as defined in section 2.1)

We impose the last restriction so that observations used in the displaced worker analysis are not used to estimate employer fixed effects. Including these workers could create a mechanical relationship between the employer fixed effects and displaced workers' earnings, hours, and wage rate losses, potentially overstating the role of employer fixed effects.

In the administrative wage records, the employer is the entity from which the state collects UI payroll taxes and to which the state "charges" UI benefits (for the purpose of experience rating the UI payroll tax). Typically, the employer is the set of establishments operating in

<sup>&</sup>lt;sup>1</sup> By removing the first quarter and the last two quarters of any worker–primary employer spell and by including at least two consecutive quarters of earnings from the same primary employer in a calendar year, we lose about 27 percent of all worker–primary employer spells (that are shorter than five consecutive quarters). If we only remove the first quarter and the last two quarters of any worker–primary employer spell (without requiring at least two consecutive quarters of earnings from the same primary employer in a calendar year), we lose about 23 percent of all worker–primary employer spells.

<sup>&</sup>lt;sup>2</sup> In Washington, 680 paid hours in approximately the previous year are required for UI eligibility.

Washington under a single owner, so for a company operating entirely in Washington (with a single or multiple addresses) the employer is a firm, and for a company with one address in Washington, the employer is also an establishment.

The first column of Table B.1 ("Full annualized panel") shows summary statistics for the annualized linked employer-employee panel—that is, after processing the quarterly records as illustrated in Figure B.1 and imposing the sample restrictions described above. The second column of Table B.1 shows the number of observations in the largest connected set, which we used to estimate the AKM model (see section 3.2 of the main text). Identification of employer fixed effects comes from workers moving between primary employers, so it is important to know how much mobility there is in the sample. Table B.1 shows that there are about 3.5 million unique workers in the largest connected set, and about a third of those workers change primary employer at least once during 2002–2014.

#### **B.2** Employer births and deaths

An employer ID may disappear if the employer becomes inactive or is acquired by or merges with another employer. In the case of a merger or acquisition, the employer's ID will change—it will appear that one employer has "died" and another has been "born." We cannot distinguish between the reorganization and the closing of an employer, but like Card, Heining, and Kline (2013), we take the view that reorganization may imply new employment policies, so it makes sense to treat reorganization as creation of a new entity and to estimate a separate (new) employer fixed effect. As Card, Heining, and Kline (2013) point out, treating assignment of a new ID to an existing employer leads to a loss of efficiency but no bias. That is, if the employer effect on wages does not change, we are estimating the same fixed effect twice, rather than once.

#### **B.3** Estimation and variance decompositions

We estimate the AKM model (equation 3 in the main text) using the linked employer-employer panel for each of the three outcomes: log earnings, log hours worked, and log wage rates, as defined earlier. Table B.2 displays the resulting variance decompositions. The variance of each outcome is decomposed five components: one due to worker effects, one due to employer effects, one due to year effects, one due to the covariance between worker and employer effects (sorting of workers and employers), and a residual. (To conserve space, we do not show the worker-year or employer-year covariances. Together, these two covariances explain about one percent of the variation in each outcome.) The numbers in italics below each variance– covariance term show the share of the total variance of each outcome attributable to that component.

Table B.2 shows that worker fixed effects explain a large share of the variation in all three outcomes: 52 percent of the variation in earnings, 45 percent of the variation in work hours, and 60 percent of the variation in hourly wage rates. This compares with worker fixed effects explaining 55 percent of the variation in earnings in Sorkin (2016) (see his Table 1.1, U.S., 2000–2008) and 51–61 percent of daily earnings in Card, Heining, and Kline (Table 3, Germany, 1985–2009).

Nevertheless, employer effects are also important. Table B.2 shows that employer fixed effects explain about 20 percent of the variation in earnings, 35 percent of the variation in work hours, and 13 percent of the variation in hourly wage rates. This compares with employer fixed effects explaining about 22 percent of the variation in earnings in Sorkin (2016) and 18–21 percent in CHK (Table 3, Germany, 1985–2009).

Table B.2 also shows that match effects—estimated from a model that regresses each outcome variable on worker–employer spell indicators and year effects—appear to be only moderately important. The adjusted– $R^2$  from the AKM model for earnings equals about 0.87 [similar to Sorkin (2016), who obtains an adjusted– $R^2$  of 0.85], whereas the adjusted– $R^2$  from a model for earnings that has a match component equals about 0.93. Although the fit is better for the match effects model, the difference between the AKM and match–effects models is not dramatic.

#### **B.4** Event studies of inter-employer mobility

Following Card, Heining, and Kline (2013), we conduct event studies of how earnings, work hours, and wage rates change when workers move between employers. The purpose of these event studies is to infer the importance of employers in determining earnings, hours, and wage rates, holding constant worker fixed effects. For example, we can follow a group of workers who start with an employer whose fixed effect ( $\psi$ ) is in the fourth quartile, and who then move to other employers. Some of these "destination" employers will have a high  $\psi$ , whereas others will have a low  $\psi$ , and observing how workers' earnings, hours, and wage rates change with these moves will provide information about employers' influence on earnings, hours, and wage rates.

The procedure for constructing these event studies is as follows. For a given outcome (earnings, hours, or wage rates) we classify employers into quartiles by their AKM–estimated employer fixed effect ( $\psi$ ). Next, for a given year *t*, we select workers in each employer quartile who have been with the employer at least two years, change employers (i.e., are observed with a different employer in year *t*+1), and remain with the subsequent employer for at least two years.

Finally, we calculate the average outcome before and after the move for each possible type of move (Q1 $\rightarrow$ Q1, Q1 $\rightarrow$ Q2, ..., Q4 $\rightarrow$ Q3, Q4 $\rightarrow$ Q4).

Figure B.2 shows the results for eight of the transitions  $(Q4\rightarrow Q4, Q4\rightarrow Q3, Q4\rightarrow Q3, Q4\rightarrow Q1, Q1\rightarrow Q4, Q1\rightarrow Q3, Q1\rightarrow Q2, Q1\rightarrow Q1)$  for log earnings. Figures B.3 and B.4 show same transitions for log work hours and log wage rates. (Table B.3 displays the data underlying these figures.)

Several points are worth noting in Figure B.2. First, workers who start with a low- $\psi$  (Q1) employer and move to a high- $\psi$  (Q4) employer experience a 70 log point increase in their earnings (and smaller effects for moves to other higher- $\psi$  employers). This change is 60 log points if adjusted by within-quartile change (see the "Adjusted change from year –2 to year 1" column in Table B.3.) Conversely, workers who start with a high- $\psi$  (Q4) employer and move to a low- $\psi$  (Q1) employer experience a 54 log point decrease in their earnings (and smaller effects for moves to other lower- $\psi$  employers). This change is 63 log points if adjusted by within-quartile change is 63 log points if adjusted by within-quartile change. In short, employers matter in determining earnings—upwardly mobile workers improve their earnings conditional on their initial quartile, and downwardly mobile workers have worse earnings conditional on their initial quartile. This tends to support the premise of the AKM approach.

Second, the pattern of movement suggests sorting of lower- $\alpha$  workers to lower- $\psi$  employers and of higher- $\alpha$  workers to higher- $\psi$  employers. For example, before the move, workers with the highest- $\psi$  employers are stratified by the type of employer to which they move—those who move to other high- $\psi$  employers have the higher pre–move earnings than moving to lower- $\psi$ employers.

Third, the average worker who moves from one Q1 employer to another Q1 employer experiences no earnings change; and similarly for workers who move Q4 $\rightarrow$ Q4, Q2 $\rightarrow$ Q2, and Q3 $\rightarrow$ Q3. The implication is that idiosyncratic matches are not of great importance; if they were, Q1 workers would be observed moving strategically to other Q1 employers—that is, moving to other Q1 employers and increasing their earnings as a result—and similarly for Q2 $\rightarrow$ Q2, Q3 $\rightarrow$ Q3, and Q4 $\rightarrow$ Q4 moves. Such within-quartile transitions provide a kind of counterfactual: workers who do not move have the same  $\alpha$  and  $\psi$ , so we would expect their profiles of earnings, work hours, and wage rates to be essentially flat. Figure B.1 suggests this is the case.

Fourth, the approximate symmetry of changes suggests that workers do not necessarily change jobs in order to obtain a pecuniary return. Rather, they may be moving to obtain some non–pecuniary benefit or compensating differential (Sorkin 2016). Fifth, all three outcome variables show slight pre- and post-transition trends, although the trends do not vary by quartile group.

# Appendix Table B.1

Summary statistics for the full annualized panel and the largest connected set (AKM analysis sample)

Full Sample	Full annualized panel	Largest connected set
Number of worker-year observations	25,578,007	22,941,274
Number of workers	4,450,785	3,508,811
Number of employers	341,553	218,593
Number of movers	1,546,094	1,463,030
Log earnings (mean)	10.321	10.432
Log hours (mean)	7.338	7.453
Log hourly wage rate (mean)	3.063	3.052

Source: Authors' tabulations of Washington administrative wage records, 2002-2014.

#### Appendix Table B.2

Variance decomposition of log earnings, log hours, and log hourly wage rates, Washington, 20022014	
(variance shares accounted for by each component in italics)	

	Var	iance of out	come and deco	AKM model fit		Match effects model fit				
	Total	Worker	Employer	Year FEs			Adj.		2	
Outcome	variance	FEs (a)	FEs (ψ)	(θ)	$2cov(\alpha,\psi)$	Residual	<u>R</u> <sup>2</sup>	RMSE	Adj. R <sup>2</sup>	RMSE
Log earnings	0.596	0.309	0.123	0.004	0.101	0.064	0.872	0.253	0.925	0.211
-		0.519	0.207	0.006	0.169	0.107				
Log hours	0.129	0.058 <i>0.449</i>	0.045 <i>0.352</i>	0.000 <i>0.001</i>	-0.013 -0.104	0.039 <i>0.303</i>	0.638	0.197	0.754	0.178
Log hourly										
wage rate	0.411	0.247 0.601	0.053 <i>0.128</i>	$0.022 \\ 0.054$	0.065 <i>0.159</i>	0.040 0.096	0.885	0.199	0.932	0.167

Source: Authors' tabulations of Washington administrative wage records, 2002-2014

*Notes:* The universe consists of workers in the AKM dataset. The decompositions include covariances between worker and employer fixed effects and year fixed effects. Because these covariances explain only about 1 percent of the variation, they are omitted from the table. The match effects model is estimated by regressing each outcome variable on worker–employer indicators and year indicators.

## Appendix Table B.3 Mean outcomes, classified by quartile of employer fixed effect

Origin/destination	Year –				Change from year -2	Adjusted change from	Number of
quartile	2	Year -1	Year 0	Year 1	to year 1	year –2 to year 1 <sup>a</sup>	observations
1 to 1	9.79	9.79	9.88	9.89	0.10	0.00	247,950
1 to 2	9.83	9.83	10.12	10.14	0.31	0.21	120,636
1 to 3	9.80	9.81	10.27	10.30	0.50	0.40	74,936
1 to 4	9.88	9.89	10.54	10.58	0.70	0.60	38,488
2 to 1	10.16	10.13	10.03	10.03	-0.12	-0.18	80,566
2 to 2	10.29	10.28	10.35	10.35	0.06	0.00	173,078
2 to 3	10.36	10.35	10.54	10.55	0.19	0.13	116,916
2 to 4	10.39	10.39	10.75	10.78	0.38	0.33	57,176
3 to 1	10.36	10.32	10.02	10.02	-0.34	-0.40	29,168
3 to 2	10.52	10.50	10.46	10.46	-0.07	-0.12	84,368
3 to 3	10.65	10.64	10.70	10.71	0.06	0.00	234,702
3 to 4	10.73	10.73	10.92	10.94	0.21	0.15	122,092
4 to 1	10.72	10.70	10.18	10.18	-0.54	-0.63	13,102
4 to 2	10.77	10.74	10.51	10.51	-0.25	-0.34	27,982
4 to 3	10.87	10.86	10.81	10.81	-0.06	-0.15	84,974
4 to 4	11.15	11.15	11.21	11.24	0.09	0.00	313,108
							1 819 242

Panel A: Mean log earnings of movers, classified by quartile of employer earnings fixed effect at origin (year = -1) and destination (year = 0) employer

*Notes:* a. Adjusted change is the change from year -2 to year 1, minus the within-quartile change from year -2 to year 1.

Panel B: Mean log hours of movers, classified by quartile of employer hours fixed effect at origin (year = -1) and destination (year = 0) employer

Origin/destination quartile	Year – 2	Year -1	Year 0	Year 1	Change from year –2 to year 1	Adjusted change from year -2 to year 1 <sup>a</sup>	Number of observations
1 to 1	7.24	7.22	7.27	7.26	0.02	0.00	221,266
1 to 2	7.27	7.24	7.49	7.47	0.20	0.18	124,596
1 to 3	7.22	7.18	7.56	7.55	0.33	0.32	69,170
1 to 4	7.21	7.17	7.65	7.64	0.43	0.42	55,296
2 to 1	7.45	7.41	7.29	7.28	-0.18	-0.17	78,030
2 to 2	7.51	7.48	7.52	7.50	-0.01	0.00	169,094
2 to 3	7.50	7.47	7.58	7.57	0.07	0.08	142,100
2 to 4	7.49	7.46	7.66	7.65	0.16	0.16	83,628
3 to 1	7.54	7.50	7.25	7.23	-0.31	-0.31	32,736
3 to 2	7.57	7.54	7.51	7.49	-0.09	-0.08	90,480
3 to 3	7.58	7.55	7.58	7.57	-0.01	0.00	189,088
3 to 4	7.59	7.57	7.66	7.65	0.05	0.06	144,280
4 to 1	7.65	7.61	7.21	7.20	-0.45	-0.44	21,302
4 to 2	7.66	7.63	7.50	7.48	-0.18	-0.17	47,296
4 to 3	7.65	7.63	7.60	7.58	-0.07	-0.07	115,634
4 to 4	7.67	7.66	7.68	7.67	0.00	0.00	235,246
							1,819,242

*Notes:* a. Adjusted change is the change from year -2 to year 1, minus the within-quartile change from year -2 to year 1.

0 0	,		0	/ 1			
Origin/destination					Change from year -2	Adjusted change from	Number of
quartile	Year –2	Year –1	Year 0	Year 1	to year 1	year –2 to year 1 <sup>a</sup>	observations
1 to 1	2.46	2.49	2.53	2.57	0.12	0.00	225,660
1 to 2	2.54	2.59	2.75	2.80	0.26	0.15	121,060
1 to 3	2.57	2.64	2.90	2.96	0.39	0.27	82,656
1 to 4	2.65	2.71	3.17	3.23	0.59	0.47	45,184
2 to 1	2.72	2.76	2.69	2.73	0.01	-0.12	97,952
2 to 2	2.86	2.90	2.95	2.99	0.13	0.00	164,396
2 to 3	2.91	2.96	3.09	3.14	0.23	0.10	127,162
2 to 4	2.93	2.98	3.27	3.33	0.40	0.27	61,112
3 to 1	2.91	2.94	2.75	2.79	-0.13	-0.27	44,052
3 to 2	3.03	3.07	3.04	3.07	0.04	-0.10	100,114
3 to 3	3.12	3.17	3.22	3.27	0.14	0.00	195,376
3 to 4	3.22	3.27	3.45	3.50	0.29	0.14	121,480
4 to 1	3.17	3.21	2.78	2.82	-0.35	-0.51	17,238
4 to 2	3.29	3.33	3.12	3.16	-0.14	-0.30	40,524
4 to 3	3.37	3.41	3.33	3.37	0.00	-0.16	95,154
4 to 4	3.59	3.64	3.68	3.75	0.16	0.00	280,122
							1,819,242

Panel C: Mean log hourly wage rate of movers, classified by quartile of employer wage fixed effect at origin (year = -1) and destination (year = 0) employer

*Notes:* a. Adjusted change is the change from year –2 to year 1, minus the within-quartile change from year –2 to year 1.

## Appendix Figure B.1 Construction of the analysis sample for the AKM dataset



*Notes*: The figure shows three hypothetical employment spells with three different employers (Er1, Er2, and Er3), each of which has the minimum five quarters required to be included in the analysis sample. The first quarter and last two quarters of each employment spell (denoted by  $\times$ ) are dropped from the analysis, and outcomes from the remaining quarters are then annualized for each calendar year, conditional on the calendar year including at least two consecutive quarters of earnings from the same primary employer. For example, outcomes for 2005 (Employment spell 1) and 2008 (Employment spell 3) are obtained by averaging the outcomes for the first, second, and third quarters of 2005 (or 2008) and multiplying by four. (The quarters used in the calculations are denoted by  $\boxdot$ .) Outcomes for 2006 (Employment spell 2) are obtained by averaging the outcomes for the third and fourth quarters of 2006 and multiplying by four. Outcomes for 2007 (part of Employment spell 2) are excluded because 2007 does not include two consecutive quarters that can be used under the selection criteria (that is, after excluding the first quarter and last two quarters of each employment spell). As a result, the data from 2007:I (denoted by  $\boxtimes$ ) are not used.

Mean log earnings of movers, classified by quartile of AKM employer earnings effects ( $\psi$ ) at origin (year = -1) and destination (year = 0) employer



Appendix Figure B.3 Mean log hours of movers, classified by quartile of AKM employer hours effects ( $\psi$ ) at origin (year = -1) and destination (year = 0) employer



Mean log hourly wage rates of movers, classified by quartile of AKM employer wage rate effects  $(\psi)$  at origin (year = -1) and destination (year = 0) employer



Estimated log earnings losses due to displacement, and displacement losses due to foregone employer fixed effects, by quintile of pre-displacement employer fixed effect for log earnings



*Notes*: The figures show estimated full earnings losses due to displacement (red circles, estimated from equation (1)) compared with displacement losses attributable to foregone employer fixed effects (blue dots, estimated from equation (4)), for workers displaced from employers grouped by quintile of their employer fixed effect for earnings. (The figure for "Employer effect Q5" is repeated from Figure 9 in the main text.) Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See sections 3.2 and 5.2 of the text for details.

Estimated log work hour losses due to displacement, and displacement losses due to foregone employer fixed effects, by pre-displacement employer fixed effect for log earnings



*Notes*: The figures show estimated full work hour losses due to displacement (red circles, estimated from equation (1)) compared with displacement losses attributable to foregone employer fixed effects (blue dots, estimated from equation (4)), for workers displaced from employers grouped by quintile of their employer fixed effect for earnings. (The figure for "Employer effect Q5" is repeated from Figure 9 in the main text.) Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See sections 3.2 and 5.2 of the text for details.

Estimated log hourly wage rate losses due to displacement, and displacement losses due to foregone employer fixed effects, by pre-displacement employer fixed effect for log earnings



*Notes*: The figures show estimated full hourly wage rate losses due to displacement (red circles, estimated from equation (1)) compared with displacement losses attributable to foregone employer fixed effects (blue dots, estimated from equation (4)), for workers displaced from employers grouped by quintile of their employer fixed effect for earnings. (The figure for "Employer effect Q5" is repeated from Figure 9 in the main text.) Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See sections 3.2 and 5.2 of the text for details.

#### Appendix C: Estimates using an alternative comparison group

The comparison group used by JLS included only workers continuously employed with their primary employer throughout the observation window (in our case, 2002–2014). As Krolikowski (2018) has pointed out, this amounts to conditioning on an outcome (employment throughout the post-displacement period), and it can be expected to lead to an overstatement of the losses of displaced workers. Accordingly, we estimate equation (1) using a different comparison group, specifically, long-tenure workers (that is, employed by the same primary employer during 2002–2007) who continued with the same employer (that is, were not displaced) during 2008–2010, but who may have changed employers or separated from the primary employer sometime after 2010. This comparison group is no longer selected on a positive outcome—continuous employment in the post-displacement period—so it is less likely than the JLS comparison group to exaggerate the long-term losses resulting from displacement. We interpret the estimates obtained using this alternative comparison group as a lower bound of the effects of displacement.

Figures C.1–C.3 show the results of estimating equation (1) for earnings, hours, and hourly wage rates, using this alternative comparison group. In the quarter after displacement, earnings drop by 41 log points (compared with 49 log point using the baseline comparison group), hours drop by 32 log points (compared with 38 log points using the baseline comparison group), and wage rates are lower by 9 log points (compared with 11 log points using the baseline comparison group). After five years, earnings are lower than the alternative comparison group's by 8 log points (compared with 15 log points using the baseline comparison group), hours are lower by 4 log points (compared with 7 log points using the baseline comparison group), and hourly wage rates are lower by 4 log points (compared with 8 log points using the baseline comparison group), and

comparison group). Hence, using the alternative comparison group suggests long-term losses that are just under half those estimated using the baseline comparison group.



Appendix Figure C.1 Earnings losses due to displacement estimated using the alternative comparison group

*Notes*: The figures show earnings losses estimated using the comparison group of non-displaced workers suggested by Krolikowski (2018)—see the discussion above. (For estimates using the baseline comparison group, see Figure 2 in the main text.) The top figure shows estimated  $\delta_k$ s—quarterly unconditional earnings lost due to displacement (in constant 2010 \$1,000s)—based on equation (1) with unconditional earnings from the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly earnings lost due to displacement—based on equation (1) with the log of earnings from the primary employer as the dependent variable. Suggested on equation (1) with the log of earnings from the primary employer as the dependent variable. Whiskers (which are very small) denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

*Source*: Authors' calculations using Washington administrative wage and claims records. See sections 2.1 and the above discussion for details of the comparison groups.

Appendix Figure C.2



Work hour losses due to displacement estimated using the alternative comparison group



Notes: The figures show quarterly work hour losses estimated using the comparison group of non-displaced workers suggested by Krolikowski (2018)-see the discussion above. (For estimates using the baseline comparison group, see Figure 3 in the main text.) The top figure shows estimated  $\delta_k$ s—quarterly unconditional hours lost due to displacement—based on equation (1) with unconditional hours from the primary employer as the dependent variable. The bottom figure shows estimated  $\delta_k$ s—logarithm of quarterly hours lost due to displacement—based on equation (1) with the log of hours from the primary employer as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement.

Source: Authors' calculations using Washington administrative wage and claims records. See sections 2.1 and the above discussion for details of the comparison groups.

Appendix Figure C.3 Hourly wage rate losses due to displacement estimated using the alternative comparison group



*Notes*: The figure show hourly wage rate losses estimated using the comparison group of nondisplaced workers suggested by Krolikowski (2018)—see the discussion above. (For estimates using the baseline comparison group, see Figure 4 in the main text.) The figure plots estimated  $\delta_k$ s—the reduction in the log hourly wage rate due to displacement—based on equation (1) with the log of the hourly wage rate at the primary employer (constant 2010 dollars per hour) as the dependent variable. Whiskers denote 95-percent confidence intervals based on standard errors clustered by worker. The vertical lines denote the quarter of displacement. *Source*: Authors' calculations using Washington administrative wage and claims records. See sections 2.1 and the above discussion for details of the comparison groups.